% rpbpubl.tex: Latex format, last revised 20210916
% Based on and supersedes plain TeX {rpbpubs.tex, rpbpubs2.tex}.
%
% To discard first page use
%
%	pdflatex rpbpubl
%	pdftk A=rpbpubl.pdf cat A2-end output t.pdf; mv t.pdf rpbpubl.pdf
% 
% (or latex rpbpubl; dvips -o -Z -p "=2" rpbpubl; ps2pdf rpbpubl.ps )
%
% To produce pdf output with type 1 fonts (so looks clear on screen):
%
%	pdflatex rpbpubl
%	xpdf rpbpubl.pdf, save pages 2-last as postscript file t.ps
%	ps2pdf12 t.ps rpbpubl.pdf
%
% Note: ORCID ID is 0000-0002-8495-7437
%
\documentclass[10pt]{article}	% Use 10pt or 12pt
\usepackage{url}
\usepackage{hyperref}
\usepackage{breakurl}
%
\oddsidemargin=1truein
\evensidemargin=1truein
\textwidth=6.3truein            % A4 paper is 8.25in wide - 0.5in
\headheight=0truein
\headsep=0truein
\topmargin=1truein
\textheight=9.5truein          
\hoffset -1truein
\voffset -1truein
%
\bibliographystyle{plain}
\begin{document}
\newcommand\lbrk{{\linebreak[0]}}
\newcommand\ibid{{\em ibid }}		% common abbreviations
\newcommand{\etal}{{\em{et~al.}}}
\newcommand\pp{{\em~pp. }}
\newcommand\rpb{{R.~P.~Brent}} 	% Richard
\newcommand\awb{{A.~W.~Bojanczyk}}	% Adam
\newcommand\sg{{Shuhong Gao}}		% Shuhong
\newcommand\fdh{{F.~R.~de~Hoog}}	% Frank
\newcommand\htg{{Hong~Tang}}		% Hong
\newcommand\glc{{G.~L.~Cohen}}		% Graeme ?
\newcommand\cwj{{C.~W.~Johnson}}	% Christopher W. Johnson
\newcommand\mhk{{M.~H.~Kahn}}		% Margaret
\newcommand\htk{{H.~T.~Kung}}		% Hsiang (not used)
\newcommand\agbl{{Alan G.~B.~Lauder}}	% Alan
\newcommand\wfl{{W.~Liang}}		% Weifa Liang
\newcommand\ftl{{F.~T.~Luk}}		% Franklin
\newcommand\bdm{{B.~D.~McKay}}		% Brendan
\newcommand\wm{{W.~Ma}}		% Wanli Ma
\newcommand\idgm{{I.~D.~G.~Macleod}}	% Iain Donald Graham
\newcommand\plm{{P.~L.~Montgomery}}	% Peter Lawrence
\newcommand\bam{{B.~A.~Murphy}}	% Brian
\newcommand\mmm{{M.~Manzur~Murshed}}	% Mohammad Manzur Murshed
\newcommand\alf{{A.~J.~van~der~Poorten}} % Alfred
\newcommand\qux{{X.~Qu}}		% Xun Qu
\newcommand\hjjr{{H.~J.~J.~te~Riele}}	% Herman
\newcommand\rbs{{R.~B.~Stanton}}	% Robin
\newcommand\wsun{{Wei~Sun}}
\newcommand\drs{{D.~R.~Sweet}}		% Doug
\newcommand\tridge{{A.~Tridgell}}	% Andrew
\newcommand\jy{{J.~Yang}}		% Jian Yang
\newcommand\jxy{{J.~X.~Yu}}		% Jeffrey Xu Yu
\newcommand\lty{{L.~T.~Yang}}		% Laurence Tianruo Yang
\newcommand\bbz{{B.~B.~Zhou}}		% Bing
\newcommand\pz{{P.~Zimmermann}}	% Paul
\newcommand\ACSCON{{Australian Computer Science Conference}} % no italics
\newcommand\ACSC{{\em Australian Computer Science Communications}}
\newcommand\ANU{{Australian National University}}
\newcommand\CACM{{\em Communications of the ACM}}
\newcommand\CMA{{Centre for Mathematical Analysis}} % Now Maths and its Applications
\newcommand\CSL{{Computer Sciences Laboratory}}
\newcommand\DCS{{Department of Computer Science}}
\newcommand\IEEETC{{\em IEEE Transactions on Computers}}
\newcommand\JACM{{\em J.~ACM}}
\newcommand\JPDC{{\em J.~Parallel and Distributed Computing}}
\newcommand\LNCS{{\em Lecture Notes in Computer Science}}
\newcommand\MC{{\em Mathematics of Computation}}
\newcommand\NM{{\em Numerische Mathematik}}
\newcommand\Proc{{Proceedings of the}}		% no italics
\newcommand\SPIE{{Society of Photo-Optical Instrumentation Engineers}} % no italics
\newcommand\SIMAX{{\em SIAM J.~Matrix Analysis and Applications}}
\newcommand\SISSC{{\em SIAM J.~Scientific and Statistical Computing}}	% pre 1993
\newcommand\SISC{{\em SIAM J.~Scientific Computing}}			% from 1993
\newcommand\TOMS{{\em ACM Transactions on Mathematical Software}}
\newcommand\TR{{Technical Report}}

\pagenumbering{roman}
\thispagestyle{empty}

\begin{thebibliography}{999}				% discard this page
\vspace*{\fill}
\centerline{\large\bf DISCARD THIS PAGE}
\vspace*{\fill}
\pagebreak[4]

\pagenumbering{arabic}					% Page 1 starts here
\thispagestyle{empty}

\centerline{\large\bf Publications of Richard P. Brent}
\vspace*{20pt}

\centerline{Most of these publications are available online from}
\centerline{\url{https://maths-people.anu.edu.au/~brent/pub/pubsall.html}.
}
\vspace*{20pt}

\bibitem{rpb001} M.~P.~C.~Legg and \rpb, Automatic contouring, 
{\em \Proc\ Fourth Australian Computer Conference}, Australian Computer
Society, Adelaide, 1969, 467--468.  CR~12\#21982.

\bibitem{rpb002} \rpb, {\em Algorithms for matrix multiplication},
Report TR-CS-70-157, DCS, Stanford (March 1970), 52\pp
% Formerly available from NTIS, \#AD705509.
%
% My scanned version is rpb002i.pdf
% Following has errors caused by scanning -> text software, so don't use.
% Available from
% \url{http://i.stanford.edu/pub/cstr/reports/cs/tr/70/157/CS-TR-70-157.pdf}.
% Combined the two versions to get rpb002.pdf
% Could not submit to arXiv as does not like pdf without Latex source.

\bibitem{rpb003} \rpb, On the addition of binary numbers,
\IEEETC\ C-19 (1970), 758--759.
CR~12\#20898.

\bibitem{rpb004} \rpb, Error analysis of algorithms for matrix multiplication
and triangular decomposition using Winograd's identity,
\NM\ 16 (1970), 145--156.
MR0279981 (43\#5702),
CR~12\#21408.

\bibitem{rpb005} \rpb, An algorithm with guaranteed convergence for finding a zero
 of a function,
 {\em {Computer J.}} 14 (1971), 422--425.
% http://www.comjnl.oupjournals.org
% http://www3.oup.co.uk/computer_journal/hdb/Volume_14/Issue_04/
 MR0339475 (49\#4234),
 Zbl~231.65046.

\bibitem{rpb006} \rpb, {\em Algorithms for finding zeros and extrema of functions
 without calculating derivatives},
 Report TR CS 198, DCS, Stanford (February 1971), 313\pp
 (Ph.~D. thesis, available from NTIS, \#AD726170.) % Is this still correct ?

\bibitem{rpb007} \rpb, A new algorithm for minimizing a function of several
 variables without calculating derivatives,
 in {\em Optimization}
 (edited by R. S. Anderssen, L.~S.~Jennings and D. M. Ryan),
 University of Queensland Press, Brisbane, 1972, 14--25.
 MR~52\#2574, 52\#9601,
 Zbl~311.90065.

\bibitem{rpb008} \rpb, On the Davidenko-Branin method for solving simultaneous
 nonlinear equations,
 {\em IBM J.~Research and Development} 16 (1972), 434--436.
 CR~14\#24419,
 MR0334498 (48\#12817),
 Zbl~276.65023.
 Also appeared as
 ``A note on the Davidenko-Branin method for solving
 simultaneous nonlinear equations'',
 Report TR RC 3506, IBM Research (August 1971), 7\pp

\bibitem{rpb009} \rpb, A modified linear scatter storage technique,
 {\em IBM Technical Disclosure Bulletin} 14, 11 (1972), 3489.

\bibitem{rpb010} \rpb, An optimal secant method for solving systems 
 of nonlinear equations,
 {\em IBM Technical Disclosure Bulletin} 15, 4 (1972), 1216--1217.

\bibitem{rpb011} \rpb, {\em Algorithms for Minimization without Derivatives},
 Prentice-Hall, Englewood Cliffs, New Jersey, 1973, 195\pp
 ISBN 0-13-022335-2.
 Reviewed in:
 {\em American Scientist} 61 (May-June 1973), 374;
 {\em Mathematical Programming} 4 (1973), 349;
 {\em {Computer J.}} 16 (1973), 314;
 \MC\ 28 (1974), 865--866;
 CR~15\#26544;
 MR0339493 (49\#4251);
 Zbl~245.65032.
 Errata: \MC\ TE 520, 29 (1975), 1166.
 MR~51\#7283.
 Reprinted January 2002 by Dover Publications, Mineola, New York.
 ISBN 0-486-41998-3.
 Reissued (with corrections) March 2013 by Dover Publications,
 Mineola, New York.  
 ISBN-13: 978-0-486-41998-5 (pbk.), 
 ISBN-10: 0-486-41998-3 (pbk.).
 Errata: \url{http://maths-people.anu.edu.au/~brent/pub/pub011.html}.\\
% To be made available in electronic edition, April 2013 (see emails 
% from John Grafton).
% MR1886182.
% http://www.doverpublications.com
% Dover version of rpb011 to LNT 20020322
A preliminary version appeared as~\cite{rpb006}.

\bibitem{rpb012} \rpb, The computational complexity of iterative methods for
 systems of nonlinear equations,
 in {\em Complexity of Computer Computations}
 (edited by R. E. Miller and J.~W. Thatcher),
 Plenum Press, New York, 1972, 61--71.
 MR~51\#9575, 52\#4703.

\bibitem{rpb013} \rpb, 
Reducing the retrieval time of scatter storage techniques,
 \CACM\ 16 (1973), 105--109.
 Zbl~251.68019.
 Also appeared as
 ``A~modified linear scatter storage technique'',
 Report TR RC 3460, IBM Research (July 1971), 20\pp
 See also ``Comment on Brent's scatter storage algorithm'',
 \CACM\ 16 (1973), 703.

\bibitem{rpb014} \rpb, Some efficient algorithms for solving systems of
 nonlinear equations,
 {\em SIAM J.~Numerical Analysis} 10 (1973), 327--344
 (George E.~Forsythe memorial issue).
 MR~48\#10096,
 CR~17\#29965,
 Zbl~258.65051.
% Available from Jstor
 A preliminary version appeared as
 {\em On maximizing the efficiency of algorithms for solving
 systems of nonlinear equations},
 Report TR RC 3725, IBM Research (February 1972), 33\pp

\bibitem{rpb015} \rpb, D. J.~Kuck and K. Maruyama, The parallel evaluation of
 arithmetic expressions without division,
 \IEEETC\ C-22 (1973), 532--534.
 MR0423869 (50\#11843).

\bibitem{rpb016} \rpb, S. Winograd and P. Wolfe, Optimal iterative processes
 for rootfinding,
 \NM\ 20 (1973), 327--341.
 CR~15\#26753,
 MR0317532 (47\#6079).
 A preliminary version appeared as
 Report TR RC 3960, IBM Research (August 1972), 29\pp

\bibitem{rpb017} \rpb, On the precision attainable with various floating-point
 number systems,
 \IEEETC\ C-22 (1973), 601--607.
 CR~14\#25960,
 Zbl~261.65036.
 Also appeared as
 Report TR RC 3751, IBM Research (February 1972), 28\pp
 arXiv:1004.3374v1

\bibitem{rpb018} \rpb, The parallel evaluation of arithmetic expressions in
 logarithmic time,
 in {\em Complexity of Sequential and Parallel Numerical Algorithms}
 (edited by J.~F.~Traub), Academic Press, New York, 1973, 83--102.
 CR~15\#26540, 15\#27335;
 MR~50\#15432;
 Zbl~273.68037.

\bibitem{rpb019} \rpb, The first occurrence of large gaps between 
successive primes,
 \MC\ 27 (1973), 959--963.
 MR~48\#8360,
 Zbl~269.10002.
% Available from Jstor
 For a sequel see~\cite{rpb057}.

\bibitem{rpb020} \rpb, Sources of error in computation,
 in {\em Error, Approximation and Accuracy}
 (edited by \fdh\ and C.~L.~Jarvis),
 University of Queensland Press, Brisbane, 1973, 122--128.
 MR~52\#2124, 54\#9069;
 Zbl~312.65035.

\bibitem{rpb021} \rpb, The distribution of small gaps between successive primes,
 \MC\ 28 (1974), 315--324.
 MR~48\#8356,
 Zbl~274.10001.
Also \MC\ UMT file, {\em ibid}, 331--332.
% Mathematics of Computation Unpublished Math Tables Collection, 1944-1994,
% Archives of American Mathematics, Center for American History, University of
% Texas at Austin. Box 97-196/17 (5663210) or 97-196/29 (5663005)
% reviewed by D. Shanks in Math. Comp. 28 (1974), 331--332 % rpb021r
% Paper and review available from Jstor
 A preliminary version appeared as
 ``Empirical evidence for a proposed distribution of
 small prime gaps'',
 Report TR CS 123, DCS, Stanford (February 1969), 18\pp

\bibitem{rpb022} \rpb, The parallel evaluation of
 general arithmetic expressions,
 \JACM\ 21 (1974), 201--206.
 CR~15\#27055,
 MR0660280 (58\#31996),
 Zbl~276.68010.
% ISI citations 282 as at 20091105

\bibitem{rpb023} \rpb, Algorithm 488: A Gaussian pseudo-random
 number generator [G5],
 \CACM\ 17 (1974), 704--706.
 Zbl~292.65003.

\bibitem{rpb024} \rpb, Irregularities in the distribution of primes and
 twin primes,
 \MC\ 29 (1975), 43--56
 (Derrick H.~Lehmer special issue).
 MR~50\#1791, 51\#5522;
 Zbl~295.10002.
% Associated tables reviewed by D. Shanks in
% Math. Comp. 29 (1975), 331               % rpb024r1
% and in Math. Comp. 30 (1976), 379.       % rpb024r2
% Paper and both reviews available from Jstor
% Tables available from UMT 1944-1994,
% see http://www.lib.utexas.edu/taro/utcah/00279/cah-00279.html
% box 97-196/17 (5663210)
 Errata: \ibid 30 (1976), 198. 		   % rpb024e
 MR~53\#302.
 See also ``Tables concerning irregularities in
 the distribution of primes and twin primes'',
 UMT 4, \ibid 29 (1975), 331;
 and ``Tables concerning irregularities in the distribution of
 primes and twin primes to $10^{11}$'',
 UMT 21, \ibid 30 (1976), 379.

\bibitem{rpb025} \rpb, {\em Numerical solution of nonlinear equations},
 DCS, Stanford (March 1975), 189\pp
 % Only Table of Contents and Outline included here.

\bibitem{rpb026} \rpb, Some high-order zero-finding methods using almost orthogonal
 polynomials,
 {\em J.~Australian Mathematical Society (Series B)}
 19 (1975), 1--29.
 MR~52\#16000,
 Zbl~333.65022.
 Also appeared as
 ``Efficient methods for finding zeros of functions whose
 derivatives are easy to evaluate'',
 \TR, DCS, CMU (December 1974), 62\pp

\bibitem{rpb027} \rpb, A class of optimal-order zero-finding methods using
 derivative evaluations,
 in {\em Analytic Computational Complexity}
 (edited by J.~F.~Traub),
 Academic Press, 1975, 59--73.
 Reviewed in {\em Australian {Computer J.}} 9 (1977), 100;
 MR~52\#15938, 54\#9073;
 Zbl~342.65030.
 Also appeared as a \TR, DCS, CMU (June 1975), 15\pp

\bibitem{rpb028} \rpb, Multiple-precision zero-finding methods and the 
 complexity of elementary function evaluation,
 in {\em Analytic Computational Complexity}
 (edited by J.~F.~Traub),
 Academic Press, New York, 1975, 151--176. % Copyright 1976
 Reviewed in {\em Australian {Computer J.}} 9 (1977), 100;
 MR0395140 (52\#15938), MR0423869 (54\#11843), 
 Zbl~342.65031.
 % Frances retyped (1999), see rpb028.*
 Also appeared as a \TR, DCS, CMU (July 1975), 26\pp % ADA014059
% http://oai.dtic.mil/oai/oai?verb=getRecord&metadataPrefix=html&identifier=ADA014059
 arXiv:1004.3412v2 % v2 corrected a small typo in v1


\bibitem{rpb029} \rpb\ and \htk, $O((n \log n)^{3/2})$ algorithms
 for composition and reversion of power series,
 in {\em Analytic Computational Complexity}
 (edited by J.~F. Traub),
 Academic Press, New York, 1975, 217--225.
 Reviewed in {\em Australian {Computer J.}} 9 (1977), 100;
 MR~52\#15938, 55\#1699;
 Zbl~342.65010.
 Also appeared as a \TR, DCS, CMU (May 1975), 7\pp

\bibitem{rpb030} J.~P. Abbott and \rpb, Fast local convergence with single and
 multistep methods for nonlinear equations,
 {\em J.~Australian Mathematical Society (Series B)}
 19 (1975), 173--199.
 MR~55\#4677;
 Zbl~359.65044, 389.65037.
 Errata: \ibid 20 (1977), 254.
 MR~58\#13673.

\bibitem{rpb031} R. S. Anderssen and \rpb\ (editors),
 {\em The Complexity of Computational Problem Solving},
 University of Queensland Press, Brisbane, 1976, 262\pp
 LC 76-374278, ISBN 0-7022-1213-X.
 Reviewed in {\em {Computer J.}} 21 (1978), 242.
 % Only first few pages included here.

\bibitem{rpb032} \rpb, The complexity of multiple-precision arithmetic,
 in~\cite{rpb031}, 126--165.
 arXiv:1004.3608v1
% Frances retyped (1999), see rpb032.*

\bibitem{rpb033} R. S. Anderssen, \rpb, D. J.~Daley and P. A. P. Moran,
Concerning\\
$\int_0^1\cdots\int_0^1({x_1}^2+\cdots+{x_k}^2)^{1/2}dx_1\ldots dx_k$
and a Taylor series method,
{\em SIAM J.\ Applied Mathematics} 30 (1976), 22--30.
MR~52\#15773,
Zbl~337.65022.
% Available from Jstor

\bibitem{rpb034} \rpb, Fast multiple-precision evaluation
 of elementary functions,
 \JACM\ 23 (1976), 242--251.
 MR0395314 (52\#16111),
 Zbl~324.65018.
% ISI citations 111 as at 20091105
% Reprinted in "Pi: A Source Book", 424-433.
 Also appeared as
 Report TR STAN-CS-75-515, DCS, Stanford (August 1975), 22\pp
 See also 
 {\em Computer Bulletin (Series 2)} 29 (September 1981), 26--27;
 and \ibid 30 (December 1981), 9--10.

\bibitem{rpb035} \rpb, {\em MP Users Guide},
 Report TR 54, Computer Centre, ANU (September 1976), 53\pp
 Revisions published as
 {\em MP Users Guide (second edition)},
 Computing Research Group, ANU (August 1978), 44{\em pp}.;
 {\em MP Users Guide (third edition)},
 TR-CS-79-08, DCS, ANU (December 1979), 73{\em pp}.;
 and {\em MP Users Guide (fourth edition)},
 TR-CS-81-08, DCS, ANU (June 1981), 73\pp
 arXiv:1004.3173v2 % pwd uutf8
 % Only fourth edition included here.

\bibitem{rpb036} \rpb, {\em Knuth's constants to 1000 decimal and 1100 octal places},
 Report TR 47, Computer Centre, ANU (September 1975), 25\pp
 See also UMT 30, \MC\ 30 (1976), 668.
% review of UMT 30 by J. W. Wrench. % rpb036r
% Tables available from UMT 1944-1994,
% see http://www.lib.utexas.edu/taro/utcah/00279/cah-00279.html
% Box 97-196/17 (5663210) or 97-196/30 (5663082)

\bibitem{rpb037} \rpb, Analysis of the binary Euclidean algorithm,
 in {\em New Directions and Recent Results in Algorithms
 and Complexity}
 (edited by J.~F.~Traub), Academic Press, New York, 1976, 321--355.
 Reviewed in {\em Australian {Computer J.}} 10 (1978), 76--77;
 MR~54\#14417, 55\#11701;
 Zbl~363.00013, 373.68040.
 Also appeared as a \TR, DCS, CMU (June 1976), 35\pp
 Extended abstract appeared in {\em SIGSAM Bulletin} (May 1976).
% NB: Some of the results are incorrect !  See Knuth, Vol. 2, 3rd edition,
%     Sec. 4.5.2, to appear about 1997.
 For errata see~\cite{rpb183}.

\bibitem{rpb038} R.~Towle and \rpb, On the time required to parse an arithmetic
 statement for parallel processing,
 in {\em \Proc\ 1976
 International Conference on Parallel Processing}
 (edited by P.H. Enslow),
 IEEE, New York, 1976, 254.
 (IEEE Catalog \#76CH1127-0C.)
 
\bibitem{rpb039} \rpb\ and \htk, Fast algorithms for composition 
 and reversion of multivariate power series, 
 in {\em Proceedings of a Conference on Theoretical Computer Science}
 held at the University of Waterloo,
 DCS, University of Waterloo, Waterloo, Ontario (August 1977), 
 149--158. 
 Zbl~404.00019, 411.68043.

\bibitem{rpb040} \rpb, Computation of the regular continued fraction
 for Euler's constant,
 \MC\ 31 (1977), 771--777.
 MR~55\#9490,
 Zbl~369.10002.
% Available from Jstor
 See also ``$\gamma$ and $\exp(\gamma)$ to 20700D and their
 regular continued fractions to 20000 partial quotients'',
 UMT 1, \ibid 32 (1978), 311. % or page 315?
% review of UMT by D. Shanks % rpb040r
 For a sequel see~\cite{rpb049}.
% Review of tables but not tables themselves included here.
% Tables available from UMT 1944-1994, box 97-196/23 (5662953)
% see http://www.lib.utexas.edu/taro/utcah/00279/cah-00279.html
% Box 97-196/23 (5662953) or 97-196/29 (5663005)

\bibitem{rpb041} A.~H.~Sameh and \rpb, Solving triangular systems
 on a parallel computer,
 {\em SIAM J.~Numerical Analysis} 14 (1977), 1101--1113.
 MR~56\#17026,
 Zbl 375.65016. % Zbl~375.65016.
% Available from Jstor
 Also appeared as Report TR UIUCDCS-R-75-766, DCS,
 University of Illinois, Urbana, Illinois (November 1975), 18\pp

\bibitem{rpb042} \rpb, A Fortran multiple-precision arithmetic package, 
 \TOMS\ 4 (1978), 57--70. 
 CR~20\#34962.
 Also appeared as a \TR, DCS, CMU (May 1976), 29\pp
 
\bibitem{rpb043} \rpb, Algorithm 524: MP, a Fortran multiple-precision
 arithmetic package [A1],
 \TOMS\ 4 (1978), 71--81.
 See also ``Remark on Algorithm 524'', \ibid 5 (1979), 518--519.
 
\bibitem{rpb044} J.~P.~Abbott and \rpb, A note on continuation methods 
 for the solution of nonlinear equations, 
 {\em J.~Australian Mathematical Society (Series B)} 
 20 (1978), 157--164.
 MR~58\#13672,
 Zbl~392.65017.

\bibitem{rpb045} \rpb\ and \htk, Fast algorithms for manipulating formal
 power series,
 \JACM\ 25 (1978), 581--595.
 CR~20\#34535,
 MR0520733 (58\#25090),
 Zbl~388.68052.
 Also appeared as a \TR, DCS, CMU (January 1976), 38\pp
 
\bibitem{rpb046} \rpb, T.~Dwyer, S.~J.~Edwards, A.~D.~Glenn, D.~A.~Hawking,
 A.~J.~Hurst, C.~W.~Johnson, N.~Justusson, T.~Kelly, B.~P.~Molinari,
 D.~L.~Poole, J.~M.~Robson and I.~R.~Simpson, 
 Comments on the draft Pascal standard, 
 \ACSC\ 1 (1979), 310--317.
 Also appeared as Report TR-CS-79-09, DCS, ANU (October 1979), 9\pp
 
\bibitem{rpb047} \rpb, On the zeros of the Riemann zeta function
 in the critical strip,
 \MC\ 33 (1979), 1361--1372.
 MR~80g:10033,
 Zbl~422.10031.
% Available from Jstor
 A preliminary version appeared as
 ``Computation of the zeros of the Riemann zeta function
 in the critical strip'', 
 Report TR CMU-CS-78-148, DCS, CMU (November 1978), 27\pp
 A progress report appeared as 
 ``The first 40,000,000 zeros of the Riemann zeta function
 lie on the critical line'', 
 {\em AMS Notices} 24 (1977), A-417, 77T-A161.
 See also RMT 11, 
 \MC\ 31 (1977), 803--804. 
 For a sequel see~\cite{rpb070}.
 
\bibitem{rpb048} \rpb\ and R.~A.~Jarvis (editors), {\em \Proc\ Third
 \ACSCON}, special issue of \ACSC\ 2, 1 (January 1980), 222\pp
 % Only first few pages included here.

\bibitem{rpb049} \rpb\ and E.~M.~McMillan, Some new algorithms for
 high-precision computation of Euler's constant,
 \MC\ 34 (1980), 305--312.
 MR~82g:10002.
% Available from Jstor
% Reprinted in "Pi: A Source Book, 2004, 448-455".
 Also appeared as Report TR LBL-8729, Lawrence Berkeley Laboratory;
 and as Report TR-CS-79-03, DCS, ANU (January 1979), 16\pp
 See also
 ``Euler's constant and its exponential to 30,100 decimals'',
 and ``The first 29,000 partial quotients in the regular continued
 fraction for Euler's constant and its exponential'',
 \MC\ UMT File.
% Tables available from UMT 1944-1994,
% see http://www.lib.utexas.edu/taro/utcah/00279/cah-00279.html
% box 97-196/28 (5663038)

\bibitem{rpb050} \rpb\ and J.~F.~Traub, On the complexity of composition
 and generalized composition of power series,
 {\em SIAM J.~Computing} 9 (1980), 54--66.
 MR~81b:68042.
 Also appeared as Report TR CMU-CS-78-128, DCS,
 CMU (May 1978), 34\pp
 Abstract appeared in {\em SIGSAM Bulletin} 12, 2 (May 1978), 9.

\bibitem{rpb051} \rpb, An improved Monte Carlo factorization algorithm,
 {\em BIT} 20 (1980), 176--184.
 MR~82a:10007,
 Zbl~439.65001.
 Also appeared as
 ``Analysis of some new cycle finding
 and factorization algorithms'',
 Report TR-CS-79-11, DCS, ANU (November 1979), 10\pp

\bibitem{rpb052} \rpb, Unrestricted algorithms for elementary and special functions,
 in {\em Information Processing 80} (edited by S.~H.~Lavington),
 North-Holland, Amsterdam, 1980, 613--619.
 CR~22\#38728,
 MR~81i:68009.
 Also appeared as Report TR-CS-79-13, DCS, ANU (November 1979), 20\pp
 arXiv 1004.3621v1 

\bibitem{rpb053} \rpb\ and \htk, The chip complexity of binary arithmetic,
 {\em \Proc\ Twelfth Annual ACM Symposium on the Theory of Computing},
 ACM, New York, 1980, 190--200.

\bibitem{rpb054} \rpb, J.~A.~Hooper and J.~M.~Yohe, An Augment interface for Brent's 
 multiple-precision arithmetic package, 
 \TOMS\ 6 (1980), 146--149.
 CR~21\#36520,
 Zbl~433.68028.
 A longer version appeared as 
 Technical Summary Report \#1868, Mathematics Research Center, 
 University of Wisconsin, Madison (August 1978), 26\pp
 
\bibitem{rpb055} \rpb\ and \htk, The area-time complexity
 of binary multiplication, 
 \JACM\ 28 (1981), 521--534. 
 CR~22\#38242,
 MR~82i:68027.
 Corrigendum: \ibid 29 (1982), 904.
 MR~83j:68046.
 Also appeared as
 Report TR-CS-79-05, DCS, ANU; 
 and as Report TR CMU-CS-79-136, DCS, CMU (July 1979), 25\pp
 
\bibitem{rpb056} \rpb\ and \htk, On the area of binary tree layouts, 
 {\em Information Processing Letters} 11 (1980), 46--48. 
 Also appeared as Report TR-CS-79-07, DCS, ANU (July 1979), 5\pp
 
\bibitem{rpb057} \rpb, The first occurrence of certain large prime gaps,
 \MC\ 35 (1980), 1435--1436.
 MR~81g:10002,
 Zbl~443.10003.
% Available from Jstor

\bibitem{rpb058} G.~M.~Baudet, \rpb\ and \htk, Parallel execution of a 
 sequence of tasks on an asynchronous multiprocessor, 
 {\em Australian {Computer J.}} 12 (1980), 105--112.
 MR~81j:68036,
 Zbl~437.68016.
 A preliminary version appeared as a
 \TR, DCS, CMU (June 1977), 28\pp

\bibitem{rpb059} \rpb, F.~G.~Gustavson and D.~Y.~Y.~Yun, Fast solution of
 Toeplitz systems of equations and computation of Pad\'e approximants,
 {\em J.~Algorithms} 1 (1980), 259--295.
 MR0604867 (82d:65033).
 Also appeared as Report TR RC 8173, IBM Research (January 1980); 
 and as Report TR-CS-79-06, DCS, ANU (June 1980), 36\pp
 
\bibitem{rpb060} \rpb\ and \htk, A regular layout for parallel adders,
 \IEEETC\ C-31 (1982), 260--264.
% ISI citations 291 as at 20091105
 MR~83b:68002.
 Also appeared as Report TR-CS-79-04, DCS, ANU; 
 and as Report TR CMU-CS-79-131, DCS, CMU (June 1979), 16\pp
 
\bibitem{rpb061} \rpb\ and J.~M.~Pollard, Factorization of the eighth Fermat
 number, \MC\ 36 (1981), 627--630.
 MR~83h:10014.
% Available from Jstor
 Also appeared as
 Report TR-CS-80-12, DCS, ANU, (September 1980), 7\pp
 Preliminary announcement in
 {\em AMS Abstracts} 1 (1980), 565, 80T-A212.
 See~\cite{rpb161} for a sequel.
% For sequel see rpb161.

\bibitem{rpb062} \rpb, {\em Topics in computational complexity
and the analysis of algorithms},
Report TR-CS-80-14, DCS, ANU, October 1980, 375\pp
(D.~Sc.~thesis, includes~\cite{rpb003,rpb012,rpb013,rpb014,rpb015,%
rpb016,rpb018,rpb022,rpb027,rpb028,rpb032,rpb034,rpb037,rpb039,rpb045,%
rpb050,rpb055,rpb060}.)
 
\bibitem{rpb063} \rpb, Comments on papers by Maddison and Westreich 
 (correspondence), 
 {\em {Computer J.}} 24 (1981), 95--96.
% http://www.comjnl.oupjournals.org
% http://www3.oup.co.uk/computer_journal
% http://www3.oup.co.uk/jnls/list/computer_journal/
%      hdb/Volume_24/Issue_01/default.html 

\bibitem{rpb064} \rpb\ and L.~M.~Goldschlager, Some area-time tradeoffs for VLSI, 
 {\em SIAM J.~on Computing} 11 (1982), 737--747.
 MR~83k:68024.
 Preliminary version appeared as
 Report \# 22, DCS, University of Queensland (August 1980), 18pp;
 and as Report TR-CS-80-15, DCS, ANU (Dec.\ 1980), 18\pp
 A shorter version appeared as~\cite{rpb072}.
 
\bibitem{rpb065} \awb, \rpb\ and \htk, Numerically stable solution
 of dense systems of linear equations using mesh-connected 
 processors, \SISSC\ 5 (1984), 95--104.
 MR~85b:65026.
 Also appeared as Report TR-CS-81-01, DCS, ANU;
 and as Report TR-CS-81-119, DCS, CMU (May 1981), 23\pp
 
\bibitem{rpb066} \rpb, Succinct proofs of primality
 for the factors of some Fermat numbers,
 \MC\ 38 (1982), 253--255.
 MR~82k:10002.
% Available from Jstor
 Also appeared as Report TR-CS-81-04, DCS, ANU, January 1981, 6\pp
 
\bibitem{rpb067} \rpb, New factors of Mersenne numbers (preliminary report), 
 {\em AMS Abstracts} 2 (1981), 367, 81T-10-246; 
 part II, \ibid 3 (1982), 132, 82T-10-22; 
 part III, \ibid 4 (1983), 197, 83T-10-138.
% Tables available from UMT 1944-1994,
% see http://www.lib.utexas.edu/taro/utcah/00279/cah-00279.html
% box 97-196/28 (5663038)
 
\bibitem{rpb068} \rpb, Efficient implementation of the first-fit strategy for
 dynamic storage allocation,
 {\em \Proc\ Fourth \ACSCON},
 special issue of \ACSC\ 3 (1981), 25--34.
 Also appeared as
 Report TR-CS-81-05, DCS, ANU, February 1981, 10\pp
 A revised version appeared as~\cite{rpb089}.

\bibitem{rpb069} \rpb, An idealist's view of semantics for integer and real types,
 {\em \Proc\ Fifth \ACSCON},
 special issue of \ACSC\ 4 (1982), 130--140.
 Also appeared as Report TR-CS-81-14, DCS, ANU, November 1981, 13\pp
 
\bibitem{rpb070} \rpb, J.~van de Lune, \hjjr\ and D.~T.~Winter,
 On the zeros of the Riemann zeta function in the critical strip, II,
 \MC\ 39 (1982), 681--688.
 MR~83m:10067.
% Available from Jstor
 Corrigendum \ibid 46 (1986), 771.
 MR~87e:11103.
 Also appeared as \TR\ NW 121/82, Mathematisch Centrum, Amsterdam;
 and as Report TR-CS-82-01, DCS, January 1982, 13\pp 
 A longer version appeared as~\cite{rpb081}.
 
\bibitem{rpb071} \rpb\ and R.R. Ewin, Design of an nMOS parallel adder, 
 {\em AUSMPC Design Community Newsletter} 2 (August 1982), 1--14. 
 A longer version appeared as 
 Report TR-CS-82-06, DCS, ANU, May 1982, 42\pp

\bibitem{rpb072} \rpb\ and L.~M.~Goldschlager, Area-time tradeoffs for VLSI circuits,
 {\em Microelectronics '82}, 
 Institution of Engineers, Australia, 
 National Conference Publication \# 82/4, April 1982, 52--56.
 See also~\cite{rpb064}.
 
\bibitem{rpb073} \rpb\ and \htk, Systolic VLSI arrays for polynomial GCD
 computation, 
 \IEEETC\ C-33 (1984), 731--736.
 Also appeared as Report TR-CS-82-05, DCS, ANU; 
 and as Report CMU-CS-82-118, DCS, CMU, May 1982, 16\pp

\bibitem{rpb074} \rpb\ and \ftl, Computing the Cholesky factorization 
 using a systolic architecture, 
 {\em \Proc\ Sixth \ACSCON},
 special issue of \ACSC\ 5 (1983), 295--302.
 Also appeared as Report TR-CS-82-08, DCS, ANU, August 1982;
 and as Report TR 82-521, DCS, Cornell University,
 September 1982, 15\pp
 
\bibitem{rpb075} \rpb\ and \ftl, {\em A systolic architecture for the
 singular value decomposition},
 Report TR-CS-82-09, DCS, ANU;
 Report TR 82-522, DCS, Cornell University;
 Report CMA-R02-82, ANU, August 1982, 12\pp
 See also~\cite{rpb084}.

\bibitem{rpb076} \rpb\ and \ftl, {\em A systolic architecture for almost 
 linear-time solution of the symmetric eigenvalue problem},
 Report TR-CS-82-10, DCS, ANU;
 Report TR 82-525, DCS, Cornell University;
 Report CMA-R03-82, CMA, ANU, August 1982, 23\pp
 See also~\cite{rpb084}.
 
\bibitem{rpb077} \rpb\ and \htk, A systolic VLSI array for integer GCD
 computation, 
 in {\em ARITH-7, \Proc\ Seventh Symposium on
 Computer Arithmetic} (edited by K. Hwang), IEEE/CS Press, 1985, 118--125.
 Also appeared as 
 Report TR-CS-82-11, DCS, ANU, December 1982 (revised April 1984);
 and as Report CMU-CS-84-135, DCS, CMU, April 1984, 33\pp
 
\bibitem{rpb078} \rpb\ and \ftl, A systolic array for the linear-time solution
 of Toeplitz systems of equations, 
 {\em J.~of VLSI and Computer Systems} 1, 1 (1983), 1--23.
% Note - lots of typesetting errors in the printed version!
 CR~8405-0339.
 Also appeared as
 Report TR 82-526, DCS, Cornell University, November 1982;
 and as Report TR-CS-83-02, DCS, ANU, January 1983, 29\pp

\bibitem{rpb079} \rpb, \htk\ and \ftl, Some linear-time algorithms for
 systolic arrays,
 in {\em Information Processing 83} (edited by R.E.A. Mason),
 North-Holland, Amsterdam, 1983, 865--876.
 Preliminary version appeared as
 Report TR-CS-82-15, DCS, ANU, December 1982;
 TR 83-541, DCS, Cornell University;
 Report CMA-R01-83, CMA, ANU, December 1982, 46\pp
 arXiv:1004.3716v1 % pwd mchhg

\bibitem{rpb080} \rpb, \ftl\ and C.~F.~Van Loan, Computation of the singular
 value decomposition using mesh-connected processors,
 {\em J.~of VLSI and Computer Systems} 1, 3 (1985), % (1983--1985),
% Vol 1, #1 was 1983 but Vol 1, #3 later.
% Must have appeared in 1985 as referred to rpb084 published in 1985.
 242--270.
 MR~86m:65033.
 Also appeared as
 Report TR 82-528, DCS, Cornell University, November 1982;
 and as Report TR-CS-83-05, DCS, ANU, January 1983, 34\pp
 
\bibitem{rpb081} \rpb, J.~van de Lune, \hjjr\ and D.~T.~Winter, The First
 200,000,001 zeros of Riemann's zeta function,
 in {\em Computational Methods in Number Theory}
 (edited by H.~W.~Lenstra,~Jr.\ and R.~Tijdeman),
 Mathematical Centre Tracts 154,
 Mathematisch Centrum, Amsterdam, 1982, 389--403.
% Preliminary version in {\em Syllabus Studieweek Getaltheorie \& Computers},
% held at the Math.\ Centrum, 1--5 Sept.\ 1980.
 MR~84h:10003, 84d:10004.

\bibitem{rpb082} \rpb\ and \htk, Systolic VLSI arrays for linear-time GCD
 computation,
 in {\em VLSI 83} (edited by F.~Anceau and E.~J.~Aas), 
 North-Holland, Amsterdam, 1983, 145--154. 
 Also appeared as Report TR-CS-83-08, DCS, ANU, May 1983, 11\pp
 
\bibitem{rpb083} \rpb, \ftl\ and C.~F.~Van~Loan, Computation of the generalized
 singular value decomposition using mesh-connected processors,
 {\em Proceedings SPIE, Volume 431, Real Time Signal Processing VI}
 (edited by Keith Bromley),
 \SPIE, Bellingham, Washington, 1983, 66--71.
 Also appeared as
 Report TR 83-563, DCS, Cornell University, July 1983; 
 and as Report CMA-R31-83, CMA, ANU, August 1983, 12\pp

\bibitem{rpb084} \rpb\ and \ftl, The solution of singular-value and symmetric
 eigenvalue problems on multiprocessor arrays,
 \SISSC\ 6 (1985), 69--84.
 MR~86i:65089.
% ISI citations 105 as at 20091105
 Also appeared as~\cite{rpb075,rpb076} and as
 Report TR 83-562, DCS, Cornell University, July 1983, 34\pp
 A preliminary version appeared in
 {\em Proceedings of a Workshop on Mathematical Programming
 and Numerical Analysis}
 (edited by S. Gustavson and R.S. Womersley),
 CMA, ANU, 1984, 38--64.
 MR~86k:65129.
 
\pagebreak[3]

\bibitem{rpb085} \rpb\ and L.~M.~Goldschlager, A parallel algorithm for context-free
 parsing,
 {\em \Proc\ Seventh \ACSCON},
 special issue of \ACSC\ 6 (1984), 7.1--7.10.
 Also appeared as Report CMA-R50-83, CMA, ANU, 1983, 11\pp

\bibitem{rpb086} \awb\ and \rpb, Tridiagonalization of a symmetric matrix
 on a square array of mesh-connected processors,
 \JPDC\ 2 (1985), 261--276.
 Also appeared as Report CMA-R45-83, CMA, ANU, December 1983, 23\pp
 
\bibitem{rpb087} \rpb\ and \ftl, The solution of singular-value problems using
 systolic arrays, 
 {\em Proceedings SPIE, Volume 495,
 Real Time Signal Processing VII}, 
 \SPIE, Bellingham, Washington, 1984, 7--12.
 Also appeared as Report CMA-R31-84, CMA, ANU, 1984;
 and as Report TR 84-626, DCS, Cornell University, August 1984, 13\pp
 
\bibitem{rpb088} \awb\ and \rpb, Parallel solution of certain Toeplitz
 least squares problems,
 {\em J.~Linear Algebra and its Applications} 77 (1986), 43--60.
 MR 88b:65160.
 Also appeared as Report CMA-R29-84, CMA, August 1984, 25\pp
 
\bibitem{rpb089} \rpb, Efficient implementation of the first-fit strategy for
 dynamic storage allocation,
 {\em ACM Trans.\ on Programming Languages and Systems}
 11, 3 (July 1989), 388--403.
% Review by G. Chandler in Computing Reviews,
% and pdf file available: see http://www.acm.org/dl
 Also appeared as Report CMA-R33-84, August 1984, 26\pp
 A preliminary version appeared as~\cite{rpb068}.

\bibitem{rpb090} \rpb, Dynamic storage allocation on a computer with
 virtual memory, 
 % Submitted to
 % {\em ACM Trans.\ on Programming Languages and Systems}
 % but needs revision (too late now ?)				<<<===
 % Also appeared as
 Report CMA-R37-84, CMA, ANU, September 1984; 
 and Report TR-CS-84-06, DCS, ANU, October 1984, 42\pp 
 A shorter version appeared as~\cite{rpb091}.
 
\bibitem{rpb091} \rpb, The most-recently-used strategy for dynamic storage
 allocation on a computer with virtual memory, 
 {\em \Proc\ Eighth \ACSCON},
 special issue of \ACSC\ 7 (1985), 17.1--17.8.
 See also~\cite{rpb090}.
 
\bibitem{rpb092} \awb, \rpb\ and \fdh, QR factorization of Toeplitz
 matrices, \NM\ 49 (1986), 81--94.
 MR0847019 (87k:65050).
 Also appeared as
 Report CMA-R07-85, CMA, ANU, May 1985, 23{\em pp}.;
 and as ``Parallel QR factorization of Toeplitz matrices'',
 {\em Proceedings SPIE, Volume 696,
 Advanced Algorithms and Architectures for Signal Processing},
 \SPIE, Bellingham, Washington, 1986.

\bibitem{rpb093} \awb, \rpb\ and \fdh, Linearly connected arrays
 for Toeplitz least squares problems,
 \JPDC\ 9 (1990), 261--270.
 Also appeared as Report CMA-R06-85, CMA, ANU, May 1985, 27\pp

\bibitem{rpb094} \rpb\ and \bdm, Determinants and ranks of random matrices
 over ${\bf{Z}}_m$,
 {\em Discrete Mathematics} 66 (1987), 35--49.
 MR~88h:15042.
 Also appeared as
 ``Determinants and ranks (mod $m$) of random integer matrices'',
 Report CMA-R25-85, CMA, ANU, August 1985, 17\pp

\bibitem{rpb095} \awb, \rpb, P.~Van~Dooren and \fdh,
 A note on downdating the Cholesky factorization,
 \SISSC\ 8 (1987), 210--221.
 MR0883767 (88d:65049).
 A modified version appeared in
 {\em Algorithms and Applications on Vector and Parallel Computers}
 (edited by \hjjr, Th.~J.~Dekker and H.~A.~van~der~Vorst),
 Elsevier, Amsterdam, 1987, 307--323.
 MR~89i:65042, 89f:65002.
 Also appeared as Report CMA-R19-85, CMA, ANU, August 1985, 20\pp

\pagebreak[3]

\bibitem{rpb096} \awb\ and \rpb,
 A systolic algorithm for extended GCD computation,
 {\em Comput.\ Math.\ Applic.\ }14 (1987), 233--238.
 MR~88m:11110.
 An earlier version appeared in
 {\em \Proc\ Ninth \ACSCON},
 special issue of \ACSC\ 8 (1986), 129--137.
 Also appeared as Report CMA-R29-85, CMA, ANU, September 1985, 15\pp

\bibitem{rpb097} \rpb,
{\em Some integer factorization algorithms using elliptic curves},
 Report CMA-R32-85, CMA, ANU,  September 1985, 20\pp
 A revision appeared as~\cite{rpb102}.	% Formerly both numbered [97]

\bibitem{rpb098} \rpb, A linear algorithm for data compression,
 {\em Australian Computer Journal} 19, 2 (May 1987), 64--68.
 Also appeared as Report TR-CS-86-10, DCS, ANU, November 1986, 9\pp
 Abstract appeared in
 {\em \Proc\ Tenth \ACSCON},
 special issue of \ACSC\ 9 (1987), 273.

\bibitem{rpb099} \bbz\ and \rpb,
 An efficient architecture for solving
 the recursive convolution equation with high throughput,
 {\em \Proc\ First IASTED International Symposium on Signal
 Processing and its Applications}
 (edited by B.~Boashash),
 Institution of Engineers, Australia, Vol. 2, 1987, 771--775.
 ISBN 0-8581-4143-4.

\bibitem{rpb100} \rpb\ and \glc, A new lower bound for odd perfect numbers,
 \MC\ 53, 187 (July 1989), 431--437.
 Supplement, {\em ibid}, S7--24.
 MR~89m:11008.
% Available from Jstor (including Supplement)
 Also appeared as Report TR-CS-88-05, CSL, ANU, February 1988, 50\pp
 For a sequel see~\cite{rpb116}.

\bibitem{rpb101} \rpb\ and \bdm, On determinants of random symmetric matrices
 over ${\bf{Z}}_m$,
 {\em Ars Combinatoria} 26A (1988), 57--64.
 MR~90g:05015, 89j:05001.
% {\em Proceedings Second International Conference on Combinatorial
% Mathematics and Computing} (edited by \bdm).
 Also appeared as Report TR-CS-88-03, CSL, ANU, February 1988, 8\pp
 arXiv:1004.5440v1	% pwd esydb
% Original LaTeX lost so retyped by Frances Page, Oxford, 1998
% rpb101tr

% Following may not have appeared so now in rpbpubs3.tex
%\bibitem{rpb102} T.~C.~Hu, K.~Moerder and \rpb,
%Graph folding and applications (extended abstract),
% a special issue of
%{\em Annals of Discrete Mathematics}, to appear.

\bibitem{rpb102} \rpb, Some integer factorization algorithms using 
elliptic curves,
{\em \Proc\ Ninth \ACSCON},
special issue of \ACSC\ 8 (1986), 149--163.
arXiv:1004.3366v1
% Submitted to arXiv, submit/0026141, 20100420 ***
% A preliminary version appeared as~\cite{rpb097}.
% Formerly both versions numbered [97].
% ACSC version retyped with minor corrections by Frances Page, Oxford, 1998
% (date 24 Sept 1985 as when submitted).
% rpb102

\bibitem{rpb103} \bbz\ and \rpb,
 A high throughput systolic implementation of the second order
 recursive filter,
 {\em Proceedings IEEE 1988 Conference
 on Acoustics, Speech and Signal Processing} % ISASSP-88
 (New York, April 1988), vol.~4, 2053--2056.
% Available from IEEE, 00197032.pdf -> rpb103i.pdf
 Also appeared as Report TR-CS-88-04, CSL, ANU,
 February 1988, 14\pp

\bibitem{rpb104} \bbz\ and \rpb,
 {\em A two-level pipelined implementation of direct-form
 recursive filters},
 Report TR-CS-88-06, CSL, ANU, April 1988, 16\pp\
 A revision appeared as~\cite{rpb105}.

\bibitem{rpb105} \rpb\ and \bbz,
 A stabilized parallel algorithm for direct-form
 recursive filters, \IEEETC\ C-40, 3 (March 1991), 333--336.
% Available from IEEE, 00076411.pdf -> rpb105i.pdf
 Also appeared as {\em A stabilized parallel implementation of
 direct-form recursive filters},
 Report TR-CS-88-07, CSL, ANU, May 1988, 9\pp

\bibitem{rpb106} \rpb, \glc\ and \hjjr, {\em An improved lower bound technique
 for odd perfect numbers}, % 10^240 version
 Report TR-CS-88-08, CSL, ANU, August 1988, 71\pp
 A revised version appeared as~\cite{rpb116}.
% There is also a preliminary version
% "A new approach to lower bounds for odd perfect numbers"
% which goes to 10^200 (first 4 pages on web site)

\bibitem{rpb107} T.~Bossomaier and \rpb, Matrix diagonalisation
by transputers,
{\em Proc.\ First Australian Transputer
and Occam Conference} (Melbourne, June 1988), 11--14.
% page numbers were 11--16, but there are only 4 pages.

\bibitem{rpb108} \rpb, Old and new algorithms for Toeplitz systems
 (keynote address),
 {\em Proceedings SPIE, Volume 975,
 Advanced Algorithms and Architectures for Signal Processing III}
 (edited by Franklin T.~Luk),
 \SPIE, Bellingham, Washington, 1989, 2--9.
 ISBN 0-8194-0010-6.
 Also appeared as Report TR-CS-88-10, ANU, July 1988, 8\pp
% Similar to #111.

\bibitem{rpb109} C.~J.~Anfinson, \rpb\ and \ftl,
 A theoretical foundation for the weighted checksum scheme,
 {\em Proceedings SPIE, Volume 975,
 Advanced Algorithms and Architectures for Signal Processing III}
 (edited by Franklin T.~Luk),
 \SPIE, Bellingham, Washington, 1989, 10--18.
 ISBN 0-8194-0010-6.
% Also submitted to special issue on systolic arrays of J. VLSI
% Signal Processing - see Frank's letter to Jose of 880929 -
% apparently rejected but may be reconsidered after revision.
% Is this the same as #112 ? 
%
% Following message from Frank 900405 - 
% The paper to the JVSP was not accepted by
% the Special Issue. But the implication ( SY Kung did not
% give an explicit answer ) was that the paper will be considered
% for a regular issue after suitable revision. This was the paper
% on the Reed-Solomon code. Another, the one to do with mod p,
% was submitted to ( and rejected by ) IEEE-TC. So now we can decide
% to freely re-submit to any journal. Or perhaps we may want
% to combine the two papers into a stronger one.

\bibitem{rpb110} \rpb, Parallel algorithms for digital signal processing,
 {\em Numerical Linear Algebra, Digital Signal Processing
 and Parallel Algorithms}
 (edited by G.~H.~Golub and P.~Van Dooren),
% Proc. NATO ASI, Leuven, Belgium, August 1988
% NATO ASI Series F: Computer and Systems Sciences, Vol. 70
 Springer-Verlag, 1991, 93--110.
% MR XN 92i:65226 (index only)
 ISBN 3-540-52300-6.
 Also appeared as Report TR-CS-89-15, DCS, ANU, June 1989, 12\pp

\bibitem{rpb111} \rpb, Parallel algorithms for Toeplitz systems,
 {\em Numerical Linear Algebra, Digital Signal Processing
 and Parallel Algorithms}
 (edited by G.~H.~Golub and P.~Van Dooren),
% Proc. NATO ASI, Leuven, Belgium, August 1988
% NATO ASI Series F: Computer and Systems Sciences, Vol. 70
 Springer-Verlag, 1991, 75--92.
% MR XN 92i:65226 (index only)
 ISBN 3-540-52300-6.
 A preliminary version appeared as~\cite{rpb108}.

\bibitem{rpb112} \rpb, \ftl\ and C.~J.~Anfinson, Checksum schemes for fault tolerant
 systolic computing, {\em Mathematics in Signal Processing II}
 (edited by J.~G.\ McWhirter), Clarendon Press, Oxford, 1990, 791--804.
 ISBN 0-19-853641-0.
% A modified version of 
% Weight selection for the checksum scheme,
% poster paper presented at the Conference on Mathematics in
% Signal Processing, Univ.\ of Warwick, UK, Dec.~1988, 8\pp

%\bibitem{rpb112.5} C.~J.~Anfinson, \rpb\ and \ftl,
% A novel fault tolerance technique for MVDR beamforming,
% or is title A theoretical foundation for the weighted checksum scheme ?
% to appear.
% submitted to {\em J.~of VLSI Signal Processing} (special issue).
% Frank wants to modify and submit to IEEE-TC special issue on Fault Tolerant
% computing (see mail 890605).

\bibitem{rpb113} \rpb, Factorization of the eleventh Fermat number
 (preliminary report),
 {\em AMS Abstracts} 10 (1989), 89T-11-73. % Check volume etc
 See also~\cite{rpb161}.
% For sequel (F10 and F11) see rpb161.

\bibitem{rpb114} \rpb, \ftl\ and C.~J.~Anfinson,
 Choosing small weights for multiple error detection,
 {\em Proceedings SPIE, Volume 1058, High Speed Computing II},
 \SPIE, Los Angeles, 1989, 130--136.
% Presented by Frank Luk at SPIE, Los Angeles, January 1989.
% Also - submitted to IEEE TC 890129 by Frank, rejected 891010

\bibitem{rpb115} \rpb, Parallel algorithms for integer factorisation,
 {\em Number Theory and Cryptography} (edited by J.~H.~Loxton),
 London Mathematical Society Lecture Note Series 154,
 Cambridge University Press, 1990, 26--37.
 ISBN {\nobreak 0-521-39877-0.}
 MR~91h:11148, 90m:11003.
 Also appeared as Report TR-CS-89-22, CSL, ANU,
 and as Report CMA-R49-89, CMA, ANU, October 1989, 12\pp

\bibitem{rpb116} \rpb, \glc\ and \hjjr, Improved techniques for lower
 bounds for odd perfect numbers, \MC\ 57 (1991), 857--868.
 MR~92c:11004.
% Available from Jstor (but I have source anyway)
% 10^300 version
 Also appeared as Report NM-R8921, Centrum voor Wiskunde
 en Informatica, Amsterdam, October 1989, 13\pp
 A longer version was submitted to the \MC\ UMT file and appeared as
 Report CMA-R50-89, CMA, ANU, October 1989, 198\pp % Check pages

\bibitem{rpb117} \rpb, {\em Factor: an integer factorization program for the
 IBM PC}, Report TR-CS-89-23, October 1989,
% Available from http://hdl.handle.net/1885/40810
 and Report CMA-R62-89, November 1989, 7\pp

\bibitem{rpb118} \rpb, Factorising, {\em Australian Math.\ Soc.\ Gazette}
 16, 5 (October 1989), 154--155.

\bibitem{rpb119} \rpb, Fast training algorithms for multi-layer
 neural nets, {\em IEEE Transactions on Neural Networks}
 2,~3 (May~1991), 346--354.
% Available from IEEE, 00097911.pdf (fee)
 Preliminary version appeared as
 Report TR-CS-90-01, CSL, ANU, January 1990, 13{\em~pp}; and as
 \TR\ NA-90-03, Department of Computer
 Science, Stanford University (1990 Forsythe Lecture~\#2), 16\pp
 Extended abstract in
 {\em \Proc\ First Australian Conference on Neural Networks}
 (edited by M.~Jabri), Electrical Engineering, University of Sydney,
 January 1990, 97--98.

\bibitem{rpb120} \rpb, Primality testing and integer factorisation,
 in {\em The Role of Mathematics in Science} (Proceedings of a Symposium
 held at the Australian Academy of Science, Canberra, 20 April 1990),
 Australian Academy of Science, 1991, 14--26.
 ISBN 0-85847-170-1.
 Also appeared as Report TR-CS-90-03, CSL, ANU, May 1990, 15\pp
% Available from http://hdl.handle.net/1885/40809

\bibitem{rpb121} P.~E.~Strazdins and \rpb, Implementing BLAS level 3 on the CAP-II,
 in~\cite{rpb123}, 121--129. % i.e. paper 12, pages 1--9.

\bibitem{rpb122} \rpb, Vector and parallel algorithms for integer factorisation,
 {\em Proceedings Third Australian Supercomputer Conference}
 (Melbourne, December 1990), Strategic Research Foundation,
 University of Melbourne, December 1990, 12\pp % no page numbers
 Also appeared as Report TR-CS-90-05, CSL, ANU, October 1990, 13\pp
% Available from http://hdl.handle.net/1885/40808
% For update see rpb193

\bibitem{rpb123} \rpb\ and M.~Ishii (editors), {\em \Proc\ First
 Fujitsu-ANU CAP Workshop}, Fujitsu Research Laboratories,
 Kawasaki, Japan, November 1990, 205\pp % estimate of pages

\bibitem{rpb124} D.~L.~Boley, \rpb, G.~H.~Golub and \ftl,
 Algorithmic fault tolerance using the Lanczos method,
 \SIMAX\ 13 (1992), 312--332.
MR~93f:65034.
% Golub 60th birthday special issue.
 Also appeared as Report TR-CS-91-03, CSL, ANU, February 1991;
% available from http://hdl.handle.net/1885/40807
 and as Report UMSI 91/94, University of Minnesota Supercomputer
 Institute, March 1991, 19\pp
% Also Stanford TR.
% Presented by Boley at the 4th SIAM Conference on Applied Linear
% Algebra, Minneapolis, Sep. 11-14, 1991

\bibitem{rpb125} \rpb, Parallel computers and parallel algorithms
 (abstract of 1990 Forsythe Lecture~\#1, also
 keynote address at the Fourteenth Australian Computer
 Science Conference), % UNSW, February 1991
 \ACSC\ 13,~1 (1991),~1.

\bibitem{rpb126} \awb, \rpb\ and \fdh, Stability analysis of fast
 Toeplitz linear system solvers,
 Report CMA-MR17-91, CMA, ANU, August 1991, 23\pp
 For a revision see~\cite{rpb144}.
% Also School of Mathematical Sciences Research Report 55.
% Version of 930121 entitled ``On stability of Toeplitz matrix
% factorization algorithms''

\bibitem{rpb127} \rpb, Computing Aurifeuillian factors,
in {\em Computational Algebra and Number Theory}
% {\em Proceedings of a Conference on Computational Algebra and Number Theory}
% held at University of Sydney, Nov.\ 1992
 (edited by W.~Bosma and A.~van der Poorten),
Mathematics and its Applications, vol.~325,
 Kluwer Academic Publishers, Boston, 1995, 201--212.
MR~96m:11111, 96c:00019.				
ISBN 0-7923-3501-5.
\url{http://doi.org/10.1007/978-94-017-1108-1_14}.
% rpb127
% Presented at Sydney CANT Conference, 18 Nov 1992.
% Introductory version of~\cite{rpb135} although published later.
% See cantams.tex, sent to Bosma 930510

\bibitem{rpb128} \rpb, Parallel algorithms in linear algebra,
 {\em Algorithms and Architectures:
 \Proc\ Second NEC Research Symposium}
 held at Tsukuba, Japan, August 1991	% invited paper
 (edited by T.~Ishiguro),
 SIAM, Philadelphia, 1993, 54--72.			
% Also edited panel discussions,
% ``Are hypercubes pass\'e'', 244--248; and
% ``Machine architecture from the view point of applications''
% (with H.~Hayashi, A.~Sangiovanni-Vincentelli and S.~A.~Ward), 257--279.
 ISBN 0-89871-312-9.
 Also Report TR-CS-91-06, CSL, ANU, August 1991, 17\pp
% Available from http://hdl.handle.net/1885/40806
 arXiv:1004.5437v1.
% Also presented as ``Linear algebra on MIMD machines'', % keynote ?
% invited paper at Australian Transputer and Occam Users Group Meeting,
% ANU, 24 September 1991.
Revision presented as keynote address at
First Fujitsu PCRF User Group Meeting, Kawasaki, Japan, Nov.\ 1992.

\bibitem{rpb129} \rpb (editor), {\em \Proc\ Second Fujitsu-ANU
CAP Workshop},
ANU, November 1991, 254\pp

\bibitem{rpb130} \rpb,
 The LINPACK benchmark on the AP~1000, {\em Proceedings of Frontiers '92}
 (McLean, Virginia, October 1992), IEEE Press, 1992, 128--135.
 ISBN~0-8186-2772-7. % IEEE Computer Society Order #2772
% Available from IEEE, 00234897.pdf -> rpb130i.pdf
 Also ``The LINPACK benchmark on the AP~1000: Preliminary report'',
 in~\cite{rpb129}, G1--G13.

\bibitem{rpb131} P.~E.~Strazdins and \rpb, The implementation of BLAS level 3 on
 the AP~1000: Preliminary report, in~\cite{rpb129}, H1--H17.

\bibitem{rpb132} \rpb, Uniform random number generators for supercomputers,
 {\em Proc.\ Fifth Australian Supercomputer Conference},
 Melbourne, December 1992, 95--104. ISBN~0-86444-270-X.
 A preliminary version appeared as
 {\em Uniform random number generators for vector and parallel
 computers}, Report TR-CS-92-02, CSL, ANU, March 1992, 16\pp\ %
% Available from http://hdl.handle.net/1885/40805
 Also Area 4 Working Note \#1.
% Based on preliminary Fujitsu "area 4" report section

\bibitem{rpb133} \rpb, On the periods of generalized Fibonacci recurrences,
\MC\ 63 (1994), 389--401.
MR~94i:11012.						
% Available from Jstor (but I have source anyway)
Also appeared as Report TR-CS-92-03, CSL, ANU, March 1992;
% Available from http://hdl.handle.net/1885/40804
Report CMA-MR8-92/SMS-31-92, Centre for Mathematics and its
Applications, April 1992, 11\pp\ %
arXiv:1004.5439v1.
Also Area 4 Working Note \#2.
% rpb133

\bibitem{rpb134} \rpb\ and \hjjr,
 Factorizations of $a^n \pm 1$, $13 \le a < 100$,
 Report NM-R9212, Centrum voor Wiskunde en Informatica, Amsterdam,
 June 1992, v$+$363\pp\ ISSN 0169-0388.
% http://oai.cwi.nl/oai/asset/5423/05423D.pdf
% rpb134u
 Also \rpb, \hjjr\ and \plm,
 Update~1 to: Factorizations of $a^n \pm 1$, $13 \le a < 100$,
 Report NM-R9419, CWI, Amsterdam,
 September 1994, iv$+$42\pp\
 %ISSN 0169-0388. % Same ISSN for all CWI reports
% rpb134u1
 Also (by the same authors)
 Factorizations of $a^n \pm 1$, $13 \le a < 100$: Update~2,
 Report NM-R9609, CWI, Amsterdam,
 March 1996, viii$+$42\pp\ %
%ISSN 0169-0388. % Same ISSN for all CWI reports
% rpb134u2
 For Update~3, see~\cite{rpb200}.

\bibitem{rpb135} \rpb, On computing factors of cyclotomic polynomials,
 \MC\ 61 (1993), 131--149
 (D.~H.~Lehmer memorial issue).
 MR1205459 (93m:11131).						
% Available from Jstor (but I have source)
 Also appeared as Reports TR-CS-92-13, CSL,
% Available from http://hdl.handle.net/1885/40797
 CMA-MR29-92 and SMS-89-92, CMA, ANU, Sept.\ 1992, 20\pp
 arXiv:1004.5466v1.
 For an introductory account, see~\cite{rpb127}.
% rpb135
% Some errata for TR
% See cyclo.tex, aurifg6.*, Gauss2.*, letters to Riesel, etc.
% See rpb127 for preliminary version (formerly rpb135pr)

\bibitem{rpb136} \rpb\ and P.~E.~Strazdins, Implementation of the BLAS level 3 and
 Linpack benchmark on the AP~1000,
 {\em Fujitsu Scientific and Technical Journal} 29, 1 (March 1993),
 61--70.							
 Also appeared as Report TR-CS-92-14, CSL, ANU, October 1992, 16\pp
% Available from http://hdl.handle.net/1885/40796
 Summary in \rpb, A.~Czezowski, P.~Price, P.~E.~Strazdins and \bbz,
 {\em Linear Algebra on the AP1000} (poster),
 First Fujitsu PCRF User Group Meeting, Kawasaki, Japan, Nov.\ 1992.

\bibitem{rpb137} \bbz\ and \rpb, Parallel implementation of eigenvalue
 algorithms on distributed memory machines,
 {\em \Proc\ 16th \ACSCON},
% (edited by G. Gupta, G. Mohay and R. Topor), Brisbane, 3-5 Feb 1993.
 special issue of \ACSC\ 15 (1993), 19--25.
 ISSN 0157-3055						

\bibitem{rpb138} \bbz\ and \rpb, 
Parallel computation of the singular value decomposition
on tree architectures,
% Old title: Parallel implementation of singular value
% 	     decomposition on tree architectures,
 {\em \Proc\ 22nd International Conference on Parallel
 Processing} (St.~Charles, Illinois, August 1993),
% 16-20 August 1993 (paper presented by E. V. Krishamurthy),
 Vol.~3, % Algorithms and Applications (S. Hariri and P. B. Berra, eds.),
 CRC Press, Ann Arbor, 1993, 128--131.			
% rpb138
 A longer version appeared as Report TR-CS-93-05, CSL, ANU,
 January 1993 (revised May 1993), 14~pp.	
% Available from http://hdl.handle.net/1885/40790	
% rpb138tr
% See also rpb153-5 for similar paper(s).

\bibitem{rpb139} \rpb, An asymptotic expansion inspired by Ramanujan,
{\em Australian Mathematical Society Gazette} 20 (December 1993), 149--155.
MR~95b:33006.
Also available as \TR\ CMA-MR02-93/SMS-10-93,
CMA, ANU, February 1993, 7~pp.
arXiv:1004.5506v1.				
% rpb139tr.*
% Submitted to Aust Math Soc Gazette 930208, accepted 930628,
% appeared December 1993 issue (received 940104).
Revision appeared as ``Ramanujan and Euler's constant'',
% presented at {\em Mathematics of Computation 1943-1993},
% Vancouver, 9-13 August 1993.	% Transparencies rpb139t.*
% Submitted abstract ``Ramanujan and Euler's constant''
{\em Proceedings of Symposia in Applied Mathematics}, Vol.~48
(edited by W.~Gautschi), American Mathematical Society, Providence,
Rhode Island, 1994, 541--545.
MR~95k:01022, MR~95j:00014.
ISBN 0-8218-0291-7. ISSN 0160-7634.	% has both ISBN and ISSN
							
% Vanc-p.tex

\bibitem{rpb140} \rpb\ and \tridge, A fast, storage-efficient parallel
sorting algorithm, {\em \Proc\ International Conference on
Application-Specific Array Processors} held at Venice, Italy, Oct.\ 1993
(edited by L.~Dadda and B.~Wah), IEEE Computer Society Press,
Los Alamitos, California, 1993, 369--379. % rpb140c
ISBN 0-8186-3492-8.					 	
% IEEE CS Press Order #3492-02, Catalog #93TH0572-8
% Available from IEEE, 00397158.pdf -> rpb140i.pdf
% rpb140 (formerly rpb140c)
Also {\em Sorting on the AP1000} (poster),
First Fujitsu PCRF User Group Meeting, Kawasaki, Japan, Nov.\ 1992.
Implementation details appeared in \tridge\ and \rpb,
{\em An implementation of a
general-purpose parallel sorting algorithm},
\TR\ TR-CS-93-01, CSL, ANU, February 1993, 24~pp.
% Available from http://hdl.handle.net/1885/40793
% rpb140tr
See also~\cite{rpb158}.

\bibitem{rpb141} \rpb\ {\em Fast normal random number generators
on vector processors,}
\TR\ TR-CS-93-04, CSL, ANU, March 1993, 6\pp\ %
% Available from http://hdl.handle.net/1885/40791
Also Area 4 Working Note \#4.					
arXiv:1004.3105v2 % pwd wsi4d
% Sent copy by e-mail to Wes Petersen, 930325
% See rpb170tr for Wallace's idea.
% rpb141tr

\bibitem{rpb142} \bbz, \rpb\ and \tridge,
 Efficient implementation of
 sorting algorithms on asynchronous distributed-memory machines,
 {\em \Proc\ 1994 International Conference on
 Parallel and Distributed Systems} (Hsinchu, Taiwan, December 1994),
% Lionel Ni, editor,
 IEEE Computer Society Press, Los Alamitos, California, 1994, 102--106.
% Available from IEEE, 00590058.pdf -> rpb142i.pdf
 ISBN 0-8186-6555-6.						
 Also \TR\ TR-CS-93-06, CSL, ANU, May 1993, 7~pp.
% Available from http://hdl.handle.net/1885/40789
% rpb142tr.tex
% Submitted to ASAP'93 (Venice) but rejected (rpb140 was accepted).
% Accepted for ICPADS'94 (Paper W435) - RPB attended and presented paper.
% Also similar paper accepted for Euromicro Workshop on Parallel and
% Distributed Processing (see rpb154) and listed in table of contents,
% but withdrawn.

\bibitem{rpb143} \awb, \rpb\ and \fdh,
Stability analysis of a general Toeplitz system solver,
{\em Numerical Algorithms} 10 (1995), 225--244.
MR~97e:65031.						
% ISSN 1017-1398
% Submitted to Numerical Algorithms (editor Brezinski) 940909.
% Accepted (letters from Brezinski, 950119 and 950330)
% Add reference to Cabay (too late).
Based on {\em Parallel algorithms and numerical stability for Toeplitz
systems} (invited paper presented by \rpb\ at			% keynote ?
SIAM Conference on Linear Algebra in Signals,
Systems and Control, Seattle, August 16--19, 1993).
Preliminary version
{\em A weakly stable algorithm for general Toeplitz systems},
\TR, TR-CS-93-15, CSL, ANU, August 1993 (revised June 1994).
% Available from http://hdl.handle.net/1885/40785
arXiv:1005.0503v1 % pwd v7hcq
% 1993 version announced but never appeared
% rpb143t, rpb143tr, rpb143, rpb143bj

\bibitem{rpb144} \awb, \rpb, \fdh\ and \drs,
On the stability of the Bareiss and related Toeplitz factorization algorithms,
{\SIMAX} 16 (1995), 40--57.
% ISSN 0895-4798
% DOI 10.1137/S0895479891221563
MR1311417 (95k:65030).							
Also \TR\ TR-CS-93-14,
CSL, ANU, November 1993, 18~pp. % rpb144tr.tex
% Available from http://hdl.handle.net/1885/40786
arXiv:1004.5510v1		% pwd 6mb7g
% A revision of rpb126 with addition of Doug Sweet as coauthor.
% RPB sent revised Latex file adam13.tex (close to rpb144.tex) to SIAM 940518.

\bibitem{rpb145} \bbz\ and \rpb,
Parallel implementation of QRD algorithms on the Fujitsu AP1000,
 {\em \Proc\ 17th \ACSCON}, % (edited by Gopal Gupta)
% Christchurch, New Zealand, Jan.~1994,
special issue of \ACSC\ 16 (1994), 733--743.
% Part C
Also \TR\ TR-CS-93-12, CSL, ANU, Nov.~1993,~11~pp.
% Available from http://hdl.handle.net/1885/40787
% rpb145tr

\bibitem{rpb146} \rpb, A.~Czezowski, M.~Hegland, P.~E.~Strazdins and B.~B.~Zhou,
Linear algebra research on the AP1000,
{\em \Proc\ Second Parallel Computing Workshop},
Fujitsu Laboratories, Kawasaki, Japan,
Nov.~1993,~P1--L1--13.				
% rpb146 (revised 931214)

\bibitem{rpb147} \rpb, {\em Uses of randomness in computation}, 
\TR\ TR-CS-94-06, CSL, ANU, June 1994, 
14~pp.\ (invited paper			% keynote ? 
presented at Theory Day, University of NSW, April 1994).
% Available from http://hdl.handle.net/1885/40780
arXiv:1004.3108v2 % pwd ttxn6
% Also RSISE Colloquium, Griffith talk.
% rpb147t, rpb147

\bibitem{rpb148} \rpb, A.~J.~Cleary, M.~Hegland, J.~H.~Jenkinson,
Z.~Leyk, M.~Nakanishi, M.~R.~Osborne, P.~J.~Price, S.~Roberts
and D.~B.~Singleton,
Implementation and performance of scalable scientific library subroutines
on Fujitsu's VPP500 parallel-vector supercomputer,
{\em \Proc\ Scalable High Performance Computing Conference},
(Knoxville, Tennessee, 23--25 May, 1994),
IEEE Computer Society Press, Los Alamitos, California, 1994, 526--533.
ISBN 0-8186-5680-8.
Also Area 4 Working Note \#16, May 1994, 9\pp\ %
							
% See e-mail 940202 from Andy Cleary
% Chris Johnson has a copy of the Proceedings.
% rpb148.dvi, rpb148.eps, rpb148.ps (no source file)

\bibitem{rpb149} \htg, \bbz, \idgm, \rpb\ and \wsun,
Parallel implementation of an adaptive 
and iterative noise % [formerly: differential iterative RMS]
filter for efficient signal restoration,
{\em Proc.\ IEEE Region~10 Ninth Annual International Conference on
Parallel Computation and Applications},
Singapore, August, 1994, Vol.~2, 672--676.		
% TENCON'94, 22-26 August 1994, Iain presented paper.
% Paper #28. See e-mail 940319.
% For source see Hong_Tang/TENCON.ps
% See also rpb151, rpb152 for two related papers.

\bibitem{rpb150} \rpb, Integer Factorization,
in {\em Grand Challenges in Supercomputing at the Australian National
University} (edited by T.~Bossomaier, D.~Singleton and M.~Kahn),
CSL, ANU, April 1994, 34--39.				
% rpb150

\bibitem{rpb151} \htg, \bbz, \idgm\ and \rpb,
A class of parallel iterative median-type algorithms
for efficient image processing,	% 8 pp
{\em \Proc\ International Conference on
Systems, Control and Information -- Methodologies and Applications (ICSCI),}
Wuhan, China, October 1994, 21--28;  also
{\em Journal of Northern Jiaotong University,}
Beijing, China, 1994.					
% Hong Tang presented paper.
% Draft 940701 from Hong Tang, uses AMSLatex ?
% Note: paper with same title accidentally sent to ICSPAT (see rpb152)	<<<===
% (see Hong_Tang/ICSPAT.ps)
% See also rpb149, rpb152
% rpb151

\bibitem{rpb152} \htg, \bbz, \idgm, \rpb\ and \wsun,
Comparisons of parallel iterative noise filters for real-time image
processing, {\em Proc.\ Fifth International Conference on Signal
Processing Applications and Technology} (ICSPAT'94),
Dallas, Texas, October 1994, Vol.~2, 1015--1020.		
% October 18-21. Hong Tang has copy of Proceedings.
% See letter of 940606 from ICSPAT
% Note: paper with same title as rpb151 accidentally sent to ICSPAT
% (see Hong_Tang/ICSPAT.ps). Hong and Iain rectified mistake ?
% See also rpb149, rpb151
% rpb152

\bibitem{rpb153} \bbz\ and \rpb, A parallel ring ordering algorithm
for efficient one-sided Jacobi SVD computations,
\JPDC\ 42 (1997), 1--10. 
% 10 pp.
% Submitted to J. Parallel & Distributed Computing
% Referees reports (received 950818) requested revision and expansion.
% Bing revised 951009.
% Accepted, see e-mail from Sahni 960821, paper S41005 (also PC961304)
% Proofs 970516, returned 970519, to appear in vol 42, 1997.
% Reprints arrived 970925, article no. PC971304
% Latest Latex file from Bing 970519, see Bing/JPDC.tex
% rpb153
% See rpb138, rpb154-5 for similar papers.
Preliminary version in
{\em \Proc\ Sixth IASTED/ ISMM International Conference on Parallel and
Distributed Computing and Systems}, Washington, D.C., October 1994,
369--372.						
% Check Proceedings						<<<===
% October 3-5, 1994. Someone from QUT presented paper ?
% PDCS'94 paper # 224-157
% see also e-mail to bing from iasted@arion.oac.uci.edu 940701.
% Is this different from ISAAC conference paper ? See Bing/ISAAC.tex

\bibitem{rpb154} \bbz\ and \rpb, On parallel implementation of the one-sided
Jacobi algorithm for singular value decompositions,
{\em Proceedings Euromicro Workshop on Parallel and
Distributed Processing}
(San Remo, Italy, January 25--27, 1995),	% Bing attended and presented paper
IEEE Computer Society Press, Los Alamitos, California, 1995, 401--408.
% Available from IEEE, 00389182.pdf -> rpb154i.pdf
% ISBN~0-8186-7031-2.
								
% See Bing/PDP.tex
% See rpb138, rpb153, rpb155 for similar papers.
% rpb154

\bibitem{rpb155} \bbz, \rpb\ and \mhk,
Efficient one-sided Jacobi algorithms for singular value decomposition
and the symmetric eigenproblem,
{\em Proc.\ IEEE First International Conference on Algorithms and
Architectures for Parallel Processing} (ICA3PP),
% Brisbane, 19-21 April, 1995. Paper number 83, session 3B
IEEE Press, 1995, 256--262. % IEEE Catalog #95TH0682-5
% Available from IEEE (fee)
ISBN 0-7803-2018-2.
% Bing/ICAPP95.tex
%								
Preliminary version appeared as	% same authors
``A one-sided Jacobi algorithm for
the symmetric eigenvalue problem'',
{\em \Proc\ Third Parallel Computing Workshop},	% PCW'94
Kawasaki, Japan, November 1994, P1-Q-1--P1-Q-7.
% Contributed paper - appeared in Proceedings, but no poster session.
% See abstract, e-mail from Bing 940920 and CWJ 941007
% Bing/PCW.tex
								
% See rpb138, rpb153-4 for similar papers.
% rpb155

\bibitem{rpb156} C.~Eldershaw and \rpb,
Factorization of large integers on some vector and parallel computers,
{\em Proceedings of Neural, Parallel and Scientific Computations}
% (edited by S. K. Aityan et al), Dynamic Publishers Inc, PO Box 48654,
% Atlanta, Georgia 30362-0654, USA.
% Proceedings of the First International Conference on Neural, Parallel
% and Scientific Computations held at Morehouse College, Atlanta, USA,
% May 28-31, 1995.
% Presented by Iain Macleod
1 (1995), 143--148.
ISBN 0-96-403989-3.
% craiga.tex
Also \TR\ TR-CS-95-01, CSL, ANU, January 1995, 6~pp.
% Available from http://hdl.handle.net/1885/40775
% rpb156tr.tex
A revision, ``Integer factorisation on the AP1000'',
appeared in {\em PCW '95: Proceedings of the Fourth International Parallel
Computing Workshop 1995},
Imperial College, London, 1995,	% 25-26 Sept 1995
233--242.                                                        
% rpb156
% RBS has copy of full Proceedings
% No editor named (John Darlington wrote Preface), vi + 382 pp, no ISSN/ISBN,
% Imperial College/Fujitsu Parallel Computing Research Centre,
% Imperial College, 180 Queen's Gate, London SW7 2BZ,
% ifpc@doc.ic.ac.uk, http://www-ala.doc.ic.ac.uk/~ifpc/
% craiga, rpb156tr (craigtr), rpb156 (craigpcw)

\bibitem{rpb157} \drs\ and \rpb,
Error analysis of a fast partial pivoting method for structured matrices,
% Old title: On the weak stability of the GKO algorithm,
% Proc.\ SPIE 40th Annual Meeting, San Diego, July 1995.
{\em Proceedings SPIE, Volume 2563, Advanced Signal Processing Algorithms},
(edited by F.~Luk), \SPIE, Bellingham, Washington, 1995, 266--280.
% DOI 10.1117/12.211404
% rpb157sp
Also \TR\ TR-CS-95-03, CSL, ANU, June 1995, 18~pp. % no "fast" in title
% Available from http://hdl.handle.net/1885/40773
arXiv:1005.0667v1 % pwd 2ewf9
% Gene suggests sending expanded version to SIMAX               
% rpb157tr
								
\bibitem{rpb158} \tridge\ and \rpb,
A general-purpose parallel sorting algorithm,
{\em International J.\ of High Speed Computing}
7 (1995), 285--301.	% No. 2, June 1995
% In ANU Library (unbound), Hancock QA76.I...
% ISSN 0129-0533
% rpb158
For a preliminary version see~\cite{rpb140}.	% Numbers split
For an extension to external sorting see
\tridge, \rpb\ and \bdm,
{\em Parallel Integer Sorting},
Internal Report, DCS, ANU, December 1995  % Report to DSTO under contract
(published as \TR\ TR-CS-97-10, CSL, ANU, May 1997), 32~pp.
% Available from http://hdl.handle.net/1885/40754
% rpb158tr

\bibitem{rpb159} \bbz\ and \rpb,
Jacobi-like algorithms for eigenvalue decomposition of a
real normal matrix using real arithmetic,
{\em \Proc\ Nineteenth \ACSCON},
% Melbourne, 31 Jan - 2 Feb 1996, editor Kotagiri Ramamohanarao,
special issue of \ACSC\ 18, 1, (1996), 367--375.
ISSN 0157-3055
% Presented by Bing
% Get final Latex file from Bing 				<<<===
% See rpb163 for a revision (very similar).
								
\bibitem{rpb160} \wm, \cwj\ and \rpb,
Concurrent programming in T-Cham,
{\em \Proc\ Nineteenth \ACSCON},
% Melbourne, 31 Jan - 2 Feb 1996, editor Kotagiri Ramamohanarao,
special issue of \ACSC\ 18, 1, (1996), 291--300.
ISSN 0157-3055
% Presented by Wanli
% Get final Latex file from Wanli				<<<===
% See also rpb164
								
\bibitem{rpb161} \rpb,
Factorization of the tenth Fermat number,
\MC\ 68 (1999), 429--451.
% number 225 (Jan.\ 1999)
MR~99e:11154.
% Note: electronic version appeared Nov. 1998.
%								
% rpb161
% 28 pp. for Math. Comp. version
% Sent to Wahlbin (for Math. Comp.) 960125, ack. 960207, ref. P-7533
% Reminder letter sent to Wahlbin 961018
% reply 961025 said still with referee
% Who is editor (not Shanks or Williams) ?
% Referee's report received 970225 (letter dated 970207),
% accepted subject to some revisions,
% sent revision to Wahlbin priority paid 970509
% accepted, scheduled for Jan 1999 issue (e-mail 970520)
% Sent "final" version Fermat\Mathcomp\F10final.tex to
% pub-submit@ams.org 970626, and also copyright form by mail.
% Page proofs received 980407, returned 980408.
% See rpb113 for announcement of F11 and rpb61 for F8
% See Fermat/rpb161t.* for Flinders 40th AustMS Conference slides
Preliminary version available as
{\em Factorization of the tenth and eleventh Fermat numbers,}
\TR\ TR-CS-96-02, CSL, ANU, Feb.\ 1996, 25\pp
% Available from http://hdl.handle.net/1885/40767
% rpb161, rpb161tr, F10*, rpb161t
								
\bibitem{rpb162} \wfl\ and \rpb,
Constructing the spanners of graphs in parallel,
{\em \Proc\ Tenth International Parallel Processing Symposium}
(IPPS~'96), IEEE/CS Press, 1996, 206--210.
ISBN 0-8186-7255-2.
% IEEE Computer Society Press Order Number PR07255
% manuscript #227G.
% Sheraton Waikiki Hotel, Honolulu, 15--19 April, 1996 (see also rpb163).
% Presented by Weifa Liang
% pdf and ps files are on IEEE CDROM in IPPS1996\papers\S05\Liang\Liang.*
% Also available from IEEE, 00508059.pdf -> rpb162i.pdf
Longer version appeared as
\TR\ TR-CS-96-01, CSL, ANU, Jan.\ 1996, 16\pp
% Available from http://hdl.handle.net/1885/40768
% For a related paper see rpb205,
%   ``Fully dynamic maintenance of $k$-connectivity in parallel''
%   by Weifa Liang, Richard P Brent and Hong Shen
% rpb162, rpb162tr
								

\bibitem{rpb163} \bbz\ and \rpb,
Jacobi-like algorithms for eigenvalue decomposition of a real normal matrix
using real arithmetic,
{\em \Proc\ Tenth International Parallel Processing Symposium}
(IPPS~'96), IEEE/CS Press, 1996, 593--600.
ISBN 0-8186-7255-2.
% IEEE Computer Society Press Order Number PR07255
% manuscript #66P.
% Sheraton Waikiki Hotel, Honolulu, 15-19 April, 1996 (see also rpb162).
% Presented by Weifa Liang, 960418.
% Get final Latex file from Bing 				<<<===
% pdf and ps files are on IEEE CDROM in IPPS1996\papers\S15\Zhou\Zhou.*
% (copied to rpb163.pdf, rpb163.ps)
% Also available from IEEE, 00508117.pdf -> rpb163i.pdf
% See rpb159 for a related paper (an earlier version).
%
% old 192 -> 163 (revised)
% Also revision ``A block Jacobi-like algorithm for eigenvalue
% decomposition of a real normal matrix using real arithmetic'',
% unpublished.
%% Submitted to \SIMAX,
%% referees' reports from Frank Luk dated 19960701,
%% needs revision by 1 June 1998 (or 1 July 1998 ?)
%% (see e-mail from O'Donnell via Bing, 980521)
%% Reference is SIMAX 029742-1
%% See Bing\SIMAX01.tex, SIMAX02.tex, sources\rpb163r.tex
%% Checked with Bing 20000323 - has not appeared because he needs to
%% revise it - now so much time has elapsed that not worth it,
%% especially as similar to [159, 163].
%% See Bing's email of 20011028 - should withdraw from SIMAX and
%% not bother to publish in a journal due to time elapsed and other
%% similar papers [159, 163 earlier version].
% Apparently Bing revised and submitted to JPDC - see~\cite{rpb207}.
For a preliminary version see~\cite{rpb159}. % Numbers split, but very similar.
								
\bibitem{rpb164} \wm, \cwj\ and \rpb,
Programming with transactions and the Chemical Abstract Machine,
{\em \Proc\ 1996 International Symposium on Parallel Architectures, Algorithms,
and Networks} (I-SPAN~'96),
IEEE Computer Society Press, Los Alamitos, California, 1996,
562--564.
% Conference held in Beijing, China, 12-14 June 1996.
ISBN 0-8186-7460-1.
% Microfiche ISBN 0-8186-7462-8
% ISSN 1087-4089
% IEEE CS Press Order Number PR07460
% Get final Latex file from Wanli				<<<===
% Available from IEEE, 00509044.pdf -> rpb164i.pdf
% See also rpb160
								

\bibitem{rpb165} \jy, \wm\ and \rpb,
From hypertext to flat text: a tool for document construction,
% AusWeb96, Landscaping the Web, edited by Roger Debreceny and Allan Ellis
{\em \Proc\ Second Australian World Wide Web Conference} (AusWeb~'96),
Southern Cross Univ.\ Press, 1996, 67--72.
ISBN 1-875855-16-5.
Also \url{http://www.scu.edu.au/ausweb96/tech/wanli}~.
% Presented by Wanli Ma at Gold Coast, 7--9 July 1996.
								
\bibitem{rpb166} \rpb, \alf\ and \hjjr,
A comparative study of algorithms for computing continued fractions of
algebraic numbers (extended abstract),
% presented at the {\em Algorithmic Number Theory Symposium II} (ANTS~II),
% Universit\'e Bordeaux~I, Talence, France, 18--23 May 1996.
% Appeared in ``Preproceedings'' (edited by Henri Cohen),
% Universit\'e Bordeaux I, May 1996, 37--50.
in {\em Algorithmic Number Theory}
(edited by Henri Cohen),
\LNCS, Vol.~1122, Springer-Verlag, Berlin, 1996, 35--47.
ISBN 3-540-61581-4.
% ISSN 0302-9743.
MR~98c:11144.
% see Herman/exstabst.tex
								
\bibitem{rpb167} \bbz, \rpb\ and \qux,
The design of a supporting environment for online parallel debugging,
{\em \Proc\ Sixth Parallel Computing Workshop},	% PCW'96
Kawasaki, Japan, November 1996, P1-S-1~--~P1-S-8.
% Paper P1-S, presented 961112 in poster session.
% rpb167
								
\bibitem{rpb168} \bbz, \rpb, \qux\ and \wfl,
A novel parallel algorithm for enumerating combinations,
{\em \Proc\ 25th International Conference on Parallel Processing},
Bloomingdale, Illinois, August 1996.
IEEE Press, Vol.~2, 70--73.
% Presented by friend of Weifa ?
% Available from IEEE (fee)
% Note letter from Ivan Stojmenovic (ivan@csi.uottawa.ca) 970107
% referring to his publications on the subject, e.g. IJCM (1992), 125.
% Bing/icpp96*
% rpb168.tex (longer unpublished version in rpb168tr.tex)
								
\bibitem{rpb169} \bbz, \rpb\ and \qux,
An efficient scheduling algorithm for multiprogramming on parallel
computing systems,
{\em \Proc\ Twentieth \ACSCON},
% (edited by M. Patel), Sydney, 5--7 February 1997,
special issue of \ACSC\ 19, 1 (1997), 336--345.
% Presented by Bing, Macquarie University, 970207.
% ISSN 0157-3055.
% Similar paper submitted to Parallel Computing conf. in Zurich, 1997 ?
% For ACSC'97 slides, see Bing\ACSC97s.ps
% rpb169
								
\bibitem{rpb170} \rpb,
{\em A fast vectorised implementation of Wallace's
normal random number generator,}
\TR\ TR-CS-97-07, CSL, ANU, April 1997, 9\pp\
% Available from http://hdl.handle.net/1885/40756
Also Area 4 Working Note \# 21.                         	
arXiv:1004.3114v1. % pwd 7zdss
% Based on {\em Fast normal random number generators for the VP and VPP},
% Fujitsu Area~4 Project Report, March 1997, 9~pp. % area-4/rann4.tex
% random/rpb170tr.tex

\bibitem{rpb171} \mmm\ and \rpb,
{\em RMSIM: a serial simulator for reconfigurable mesh parallel computers},
\TR\ TR-CS-97-06, CSL, ANU, April, 1997, 10\pp\ % + 2 front-matter
% Available from http://hdl.handle.net/1885/40757

\bibitem{rpb172} \qux, \jxy\ and \rpb,
A mobile TCP socket,
{\em Proc.\ IASTED International Conference on Software Engineering}
(San Francisco, Nov.~1997), 5~pp. % 2-5 Nov., regular (5 page) paper
% page numbers ?
% International Association of Science and Technology
% for Development - IASTED
% Paper SE18, see letter from Roger Lee, 970819
% or #276-111, see letter from Pen Harms 970829 for the program committee
A longer version appeared
as \TR\ TR-CS-97-08, CSL, ANU, April 1997, 24\pp\
% Available from http://hdl.handle.net/1885/40885
% rpb172tr
% Shorter version submitted to MobiCom '97

\bibitem{rpb173} \rpb,
Numerical stability of some fast algorithms for structured matrices,
% (invited paper)
{\em \Proc\ of the Sixth Workshop on Scientific Computing} % WSC'97
(Hong Kong, 10--12 March 1997),
% edited by Gene H. Golub, Shui-Hong Lui, Franklin T. Luk and
% Robert J. Plemmons
Springer-Verlag, 1998, 41--48.
ISBN 981-3083-60-3. % (paperback)
% For extended version see rpb177
% rpb173, HK_paper
						
\bibitem{rpb174} \mmm\ and \rpb,
Constant time algorithms for computing the contour of maximal elements
on the reconfigurable mesh,
% Special issue % on Bus-Based Architectures of
{\em Parallel Processing Letters} 8 (1998), 351--361.
% See e-mail from Yi Pan 971203. I have some reprints.
% rpb174.ps
Also appeared as
\TR\ TR-CS-97-09, CSL, ANU, May 1997, 9\pp\
% Available from http://hdl.handle.net/1885/40755
% rpb174tr.ps
Shorter version in
{\em Proc.\ 1997
International Conference on Parallel and Distributed
Systems} (ICPADS'97, Seoul, Korea, Dec.\ 1997),
IEEE Computer Society Press, Los Alamitos, California, 1997, 172--177.
% 10-13 Dec, 1997, presented by Murshed who has a copy of Proceedings
ISBN 0-8186-8227-2.
% IEEE CS Order Number PR08227
% Available from IEEE, 00652596.pdf -> rpb174i.pdf
% Manzur has original.
% rpb174pr.ps
								
\bibitem{rpb175} \rpb, R.~E.~Crandall, % not R. S. Crandall
K.~Dilcher and C. Van Halewyn,
Three new factors of Fermat numbers,
\MC\ 69 (2000), 1297--1304. % July 2000
MR~2000j:11194.
%\hbox{S~0025-5718(00)01207-2}
%(published electronically 1 March 2000).
For a preliminary version see
\rpb, R.~E.~Crandall and K.~Dilcher,
{\em Two new factors of Fermat numbers,}
\TR\ TR-CS-97-11, CSL, ANU, May 1997, 7\pp\
% Available from http://hdl.handle.net/1885/40753
% Has factors of F13 and F16
% See Fermat\BCD.tex, submitted to Math. Comp. by Dilcher about 970526.
% Withdrawn and revised, now Fermat\BCDH.tex, includes factor of F15,
% resubmitted by Dilcher about 970717
% Accepted (P-7974) according to email from Hugh Williams 990401
% tex file from Dilcher (Fermat/BCDH.tex) 990428
% rpb175tr, rpb175[=BCDH]

\bibitem{rpb176} \mmm\ and \rpb,
Algorithms for optimal self-simulation of some restricted reconfigurable
meshes,
{\em Proc.\ Second International Conference on Computational Intelligence and
Multimedia Applications} (ICCIMA'98, Monash Univ.,	% Churchill campus
Feb.\ 1998), % 9-11 Feb.
% edited by Henry Selvaraj and Brijesh Verma
World Scientific, Singapore, 1998, 734--744.
% Presented by Manzur
ISBN 981-02-3352-3.
% rpb176.ps
Longer version appeared as % (longer version)
\TR\ TR-CS-97-16, CSL, ANU, July 1997, 12\pp\	% 970715
% Available from http://hdl.handle.net/1885/40748
% Manzur has original.
% rpb176tr
							
\bibitem{rpb177} \rpb,
Stability of fast algorithms for structured linear systems,
in {\em Fast Reliable Algorithms for Matrices with Structure}
(editors, Ali H.\ Sayed and Thomas Kailath), SIAM, Philadelphia, 1999,
103--116. % Chapter 4, pp.\ 103--116.
% Proofs from Sayed 980401, returned about 980413.
% Final proofs from SIAM 990409, return to Sayed by 990503
% received free copy 991011
% member price $US47.50, code OT67
ISBN 0-89871-431-1.
% https://doi.org/10.1137/1.9781611971354.ch4
% For preliminary version see rpb173
Preliminary version available as \TR\ TR-CS-97-18,
CSL, ANU, September 1997, 13\pp\
% Available from http://hdl.handle.net/1885/40746
arXiv:1005.0671v1
% For shorter version see rpb173
% rpb177, rpb177tr, siambook

\bibitem{rpb178} \bam\ and \rpb,
On quadratic polynomials for the number field sieve,
{\em Proc.\ Fourth Australasian Theory Symposium}
(CATS'98, Perth, Feb.\ 1998),	% 2-6 Feb
% Xuemin Lin, ed., Springer-Verlag Singapore,
special issue of \ACSC\ 20, 3 (1998), 199--215.
ISBN 981-3083-92-1.
MR~2000i:11189.
% rpb178
Longer version available as
\TR\ TR-CS-97-17, CSL, ANU, August 1997, 18\pp\
% Available from http://hdl.handle.net/1885/40747
% rpb178tr
								
\bibitem{rpb179} \qux, \jxy\ and \rpb,
Implementation of a portable-IP system for mobile TCP/IP,
{\em Proc.\ 21st Australasian Computer Science Conference}
(ACSC'98, Perth, Feb.\ 1998),
% Chris McDonald, ed., Springer-Verlag Singapore,
special issue of \ACSC\ 20, 1 (1998), 499--510.
ISBN 981-3083-90-5.
Also appeared as \TR\ TR-CS-97-19, CSL, ANU, November 1997, 11\pp\
% Available from http://hdl.handle.net/1885/40745
% rpb179tr

\bibitem{rpb180} \bbz, \qux\ and \rpb,
Effective scheduling in a mixed parallel and sequential computing
environment, {\em Proc.\ 6th Euromicro Workshop on Parallel and
Distributed Computing}, Madrid, Jan.\ 1998, 32--37.
% Available from IEEE, 00647176.pdf -> rpb180i.pdf
% rpb180
							
\bibitem{rpb181} \bbz, \rpb, D.~Walsh and K.~Suzaki,
Job scheduling strategies for networks of workstations, in
{\em Job Scheduling Strategies for Parallel Processing} 
(Proc.\ Fourth JSSPP Workshop, Orlando, Florida, March 1998),
% G. Feitelson and L. Rudolph (editors),
\LNCS, Vol.~1459,
Springer-Verlag, Berlin, 1998, 143--157.
% Get final LaTeX version from Bing	
% See e-mail from Feitelson 980206 via Bing
% Details at http://www.cs.huji.ac.il/~feit/parsched/
% rpb181 
								
% I saw draft 971209, 14 pp.
% New version for ICS98, title changed (not accepted)

\bibitem{rpb182} \rpb, L.~Grosz, D.~L.~Harrar~II, M.~Hegland,
M.~H.~Kahn, G.~Keating, G.~J.~Mercer,
M.~Nakanishi,
O.~Nielsen, M.~R.~Osborne,
and B.~B.~Zhou,
Development of a mathematical subroutine library for Fujitsu
vector parallel processors,
{\em Proc.\ ACM International Conference on Supercomputing} (ICS'98),
Melbourne, July 1998, 13--20.
% Melbourne, 13--17 July 1998.
% pdf file available from http://www.acm.org/dl
% https://dl.acm.org/doi/10.1145/277830.277837
% For similar but longer paper see 187
								
\bibitem{rpb183} \rpb,
Twenty years' analysis of the binary Euclidean algorithm,
in {\em Millennial Perspectives in Computer Science: Proceedings of the
1999 Oxford~-- Microsoft Symposium in honour of Sir Tony Hoare}
(edited by J.~Davies, A.~W.~Roscoe and J.~Woodcock),
Palgrave, New York, 2000, 41--53.	% Palgrave is part of MacMillan
ISBN 0-333-92230-1.
% rpb183, bingcd/hoare/hoaresym
Extended version available as
{\em Further analysis of the Binary Euclidean algorithm},
Report PRG-TR-7-99, Oxford University Computing Laboratory,
November 1999, 18~pp.
arXiv:1303.2772v1 % 12 Mar 2013
% rpb183tr
% Related talks at AofA'99, Barcelona, 14--18 June 1999.
% See bingcd\fun98\fun98abs.* and slides\Warwickt.*, sources\aofa99t.*
% also series99\lec05
% Corrects some errors in~\cite{rpb037}.
								
\bibitem{rpb184} \mmm\ and \rpb,
Serial simulation of the % "the" missing in title
reconfigurable mesh, an image understanding architecture,
in {\em Advances in Computer Cybernetics (Volume~V),
Proc.\ International Symposium on Audio, Video, Image Processing
and Intelligent Applications} (ISAVIIA'98,
Baden-Baden, Germany, 1998), % 17-21 August 1998
% The International Institute for Advanced Studies in Systems Research
% and Cybernetics
% edited by S. M. Rahman and G. E. Lasker,
IIASSRC, Windsor, Canada, 1998, 92--97.
% IIAS-67-98,
% ISBN 0921836511
% Presented by Dr Syed Mahbubur Rahman (Monash, Churchill Campus)
% rpb184.ps
								
\bibitem{rpb185} \rpb,
Random number generation and simulation on vector and parallel computers
(extended abstract),
% invited paper for Euro-Par'98 (Southampton, 1-4 Sept 1998).
{\em Proc.\ Fourth International Euro-Par Conference}
(Southampton, UK, 1--4 Sept 1998),
% D. Pritchard and J. Reeve (editors),
\LNCS, Vol.~1470, Springer-Verlag, Berlin, 1998, 1--20.
ISBN 3-540-64952-2, ISSN 0302-9743.
% See {\tt http://www.springer.de/comp/lncs}
% Sent to Southampton ftp address 980525
% Proceedings in OUCL library
% See Europar/brent.tex, rpb185.tex
% rpb185
								
\bibitem{rpb186} \mmm\ and \rpb,
Adaptive $AT^2$ optimal algorithms on reconfigurable meshes,
{\em Proc.\ Tenth IASTED Internat.\ Conf.\ on Parallel and Distributed
Computing and Systems},
Las Vegas, Nevada, Oct.\ 1998, 190--195.	% 28-31 Oct
% rpb186pr.ps
Longer version available as \TR\ TR-CS-98-02,
CSL, ANU, March 1998, 15\pp\
% Available from http://hdl.handle.net/1885/40822
% rpb186tr.ps
Revision appeared in {\em Parallel Computing} 26 (2000), 1447--1458.
% Reprint received from Manzur 20010213.
% Available from http://www.sciencedirect.com
% rpb186.ps
								
\bibitem{rpb187} \rpb, L.~Grosz, D.~L.~Harrar~II, M.~Hegland,
M.~H.~Kahn, G.~Keating, G.~J.~Mercer,
M.~Nakanishi, M.~R.~Osborne, and B.~B.~Zhou,
Design of the scientific subroutine library for the Fujitsu VPP300,
{\em Proc.\ Third High Performance Computing Asia Conference and
Exhibition}, Singapore, Sept.\ 1998, 424--438. % 21-25 Sept
% expanded version of 182 (but excludes RNG)
							
\bibitem{rpb188} \rpb,
Computer arithmetic~-- a programmer's perspective
(keynote address), abstract in
{\em Proc.\ 14th IEEE Symposium on Computer Arithmetic}
(I.~Koren and P.~Kornerup, eds.),
IEEE Computer Society, Los Alamitos, California, 1999, 2.
% Conference held in Adelaide, 14--16 April, 1999
ISBN 0-7695-0116-8, 0-7695-0118-4 (microfiche).
% IEEE Computer Society Order Number PR0116
% IEEE Order Plan Catalog Number 99CB36336
% ISSN 1063-6889
% arith14t
							
\bibitem{rpb189} \bbz, \rpb, \cwj\ and D.~Walsh,
Job re-packing for enhancing the performance of gang scheduling, in
{\em Job Scheduling Strategies for Parallel Processing}
(Proc.\ Fifth JSSPP Workshop, San Juan, Puerto Rico, April 1999), 
% IPPS/SPDP'99.
\LNCS, Vol.~1659, Springer-Verlag, Berlin, 1999, 129--143. 
% Available from
% http://www.cs.huji.ac.il/~feit/parsched/parsched99.html#prog
%   Springer-Verlag LNCS Vol 1659
%   Title:   Job Scheduling Strategies for Parallel Processing
%   Editors: Dror G. Feitelson and Larry Rudolph
%   ISBN:    3-540-66676-1
% For details see http://www.ippsxx.org and email from Bing 990321, 990512
% Paper presented by Bing (funded by RPB's CAP Q-fund ?)
% Bing/job-repacking.*, rpb189.ps

\bibitem{rpb190} \mmm\ and \rpb,
Maximal contour algorithms on constrained reconfigurable meshes,
{\em Proc.\ 1999 International Conference on Parallel and Distributed
Processing Techniques and Applications}
(PDPTA'99, Las Vegas, Nevada, 28~June~-- 1~July, 1999),
% Hamid R.\ Arabnia, editor,
CSREA Press, Vol.~4, 1999, 2238--2244.
ISBN 1-892512-12-2, 1-892512-15-7.
% Paper #205P (see emails 990315, 990419).
% Student rego is $US340 with refund of $US150 if can not attend;
% paper will still be published in Proceedings (?).
% I paid the $US340 for Manzur (on BB acct)
% Refund of $US150 (cheque to Manzur) 990906 - forget the difference
% Manzur sent photocopy of paper from Proceedings, 20000110
% rpb190.ps

\bibitem{rpb191} \mmm\ and \rpb,
A new adaptive sorting algorithm on the reconfigurable mesh,
an image understanding architecture,
in {\em Advances in Intelligent Computation and Multimedia Systems}
(edited by S.~Rahman, O.~Baiocchi and G.~Lasker),
IIASSRC, % International Institute for Advanced Studies in Systems Research
	 % and Cybernetics
Windsor, Canada,
1999, 183--188.
ISBN 0-921836-80-5.
% Proc.\ ISIMADE'99
% International Symposium on Intelligent Multimedia
% and Distance Education, in conjunction with 11th International Conference
% on Systems Research, Informatics and Cybernetics,
% held at Baden-Baden, Germany, 2--7 August 1999.
% Paper #I581
% Rego is $US335 plus $US30 for shipping Proceedings
% No speaker but will appear in Proceedings.
% I paid for Manzur (on BB acct) - claimed from Oxford 991118 (pounds) 224

\bibitem{rpb192} \bbz\ and \rpb,
% For old 192, see 163 (revised) !
Gang scheduling with a queue for large jobs,
{\em Proc.\ 15th International Parallel and Distributed Processing Symposium,}
San Francisco, April 2001, 8\pp	% no page numbers
% Available from IEEE, 00925002.pdf -> rpb192i.pdf
% IEEE reference 0-7965-0990-8/01
% rpb192 to LNT 20020322
% rpb192, IPDPS2001

\bibitem{rpb193} \rpb,
Some parallel algorithms for integer factorisation,
% invited paper for Euro-Par'99 (Toulouse, 1-3 Sept 1999).
{\em Proc.\ Fifth International Euro-Par Conference}
(Toulouse, France, 1--3 Sept 1999),
% edited by P.~Amestoy {\em et al},
%% P. Amestoy, P. Berger, M. Dayd\'e, I. Duff, V. Frayss\'e,
%% L. Giroud and D. Ruiz
\LNCS, Vol.~1685, Springer-Verlag, Berlin, 1999, 1--22.
% Updated version of rpb122
% Sent source by e-mail to Iain Duff 990603,
% see Europar/1999/rpb193sp.tex
% See {\tt http://www.springer.de/comp/lncs} for series
ISBN 3-540-66443-2, ISSN 0302-9743.
% Note that the ``official'' LNCS version rpb193sp.pdf does not have my
% predictions about future ECM records; these are in the ``extended and
% updated'' version rpb193.pdf
% rpb193
								
\bibitem{rpb194} \bbz, P.~Mackerras, \cwj, D.~Walsh and \rpb,
An efficient resource allocation scheme for gang scheduling,
{\em Proc.\ IEEE International Workshop on Cluster Computing}
(Melbourne, 2--3 December 1999), 187--194.
% IWCC'99
% Available from IEEE, 00810824.pdf -> rpb194i.pdf
% Similar to 198, but different authors !
% Bing/iwcc99.ps.gz, rpb194.ps.gz, rpb194.pdf (from IEEE)
								

\bibitem{rpb195} \mmm\ and \rpb,
How promising is the $k$-constrained reconfigurable mesh, %~?,
{\em Proc.\ 15th ISCA Int.\ Conf.\ on Computers and their Applications}
% (CATA'2000),
(New Orleans, 29--31 March 2000), 9~pp.			% page numbers ?
% Manzur presented it.
% See emails from Manzur 19991111, 20000210 and ps/Murshed/rest_gen.ps
								
\bibitem{rpb196} \rpb,
Recent progress and prospects for 
integer factorisation algorithms (keynote address),
{\em Computing and Combinatorics:
Proc.\ Sixth Annual International Computing and Combinatorics Conference}
%(COCOON 2000)
(Sydney, July 26--28, 2000),
% edited by D.-Z.~Du, P.~Eades,
% V.~Estivill-Castro, X.~Lin and A.~Sharma,
\LNCS, Vol.~1858, Springer-Verlag, Berlin, 2000, 3--22.
% ISSN 0302-9743
ISBN 3-540-67787-9.
MR~2002h:11138.
% rpb196
Preliminary version available as Report PRG-TR-03-00, Oxford University
Computing Laboratory, 26 April 2000, 17~pp.
% rpb196tr

\bibitem{rpb197} \rpb,
{\em Public key cryptography with a group of unknown order},
Report PRG-TR-02-00, Oxford University Computing Laboratory,
5 June 2000, 11~pp.
% ecc/eccsig, rpb197tr

\bibitem{rpb198} \bbz, D.~Walsh and \rpb,
Resource allocation schemes for gang scheduling, in
{\em Job Scheduling Strategies for Parallel Processing}
(Proc.\ Sixth JSSPP Workshop, Cancun, Mexico, May 2000),
% (edited by D.~G.~Feitelson and L.~Rudolph),
\LNCS, Vol.~1911, Springer-Verlag, Berlin, 2000, 74--86.
% Similar to 194, but different authors !
% Bing/WJSSPP00/*.ps.gz
% rpb198.pdf
								
\bibitem{rpb199} \rpb, S.~Larvala and \pz,
A fast algorithm for testing reducibility of trinomials mod~2
and some new primitive trinomials of degree 3021377,
\MC\ 72 (2003), 1443--1452.
MR~2004b:11161. % MR1972745 (2004b:11161).
% Published electronically 20021218.
% Submitted to Math. Comp. 20010627; ack. 20010712, ref. P-8926
% Accepted by letter of 20011116 from Wahlbin
% and finally accepted by email of 20020113 from Chi-Wang Shu.
% Sent Latex file to AMS 20020206 (ref MCOM1478)
% rpb199.*
Also {``A fast algorithm for testing irreducibility of trinomials mod~2
(preliminary report)''},
Report PRG TR-13-00, 30~December 2000. % Revised 28 Feb 2001 !
% rpb199tr
% This should be rpb199pr.* so revision can be rpb199tr.* 	<<<===

\bibitem{rpb200} \rpb, \plm\ and \hjjr,
Factorizations of Cunningham numbers with bases 13 to 99,
Report PRG TR-14-00, 31~December 2000, vii$+$502\pp\
% There are two versions - the later one (rpb200) has different pagination
% and a few errors in text of earlier one (rpb200tr) corrected.
arXiv:1004.3169v2. % pwd 3ex79
Abridged version 
appeared as Report MAS-R0107, CWI, Amsterdam, July 2001, 28\pp\
% http://www.cwi.nl/static/publications/reports/abs/MAS-R0107.html
% rpb200tr, rpb200, rpb200CWItr.ps
% rpb200 CWI report to LNT 20020322

\bibitem{rpb201} \rpb, \sg\ and \agbl,
Random Krylov spaces over finite fields,
{\em SIAM J.\ on Discrete Mathematics} 16 (2003), 276--287.  % #2
MR~2004c:11231. % MR~1~982~139.
% #0393881, submitted August 2001,
% tentatively accepted 20021029, revision sent 20021105.
% preprint, August 2001, 17\pp\
% see Lauder/KryAug*.*
% for final version see Lauder/SIAM/39388.pdf,.tex from SIAM 
% published electronically about 20030530
% rpb201 preprint to LNT 20030321
% rpb201

\bibitem{rpb202} \bbz\ and \rpb,
On the development of an efficient coscheduling system, in
{\em Job Scheduling Strategies for Parallel Processing}
(Proc.\ Seventh JSSPP Workshop, MIT, Cambridge, Mass., June 2001),
% edited by D.~G.~Feitelson and L.~Rudolph,
\LNCS, Vol.~2221, Springer-Verlag, Berlin, 2001, 103--115.
% WJSSPP01. Presented by Bing.
% rpb202 to LNT 20020322
%% rpb202.ps.gz, rpb202.tex

\bibitem{rpb203} \lty\ and \rpb,	% formerly 206
% [79] of Laurence's transfer report, May 2003
Quantitative performance analysis of the improved quasi-minimal residual method
on massively distributed memory computers,
{\em Advances in Engineering Software} 33 (2002), 169--177.
% 33, 3 (March 2002), 169--177.
% Received 19970701, revised 20010309, accepted 20010629
% See www.elsevier.com/locate/advengsoft
% ps/Yang/aes.pdf (revised Nov 2001) not quite final version
% Note that my final corrections did not make it in time for the paper.
% Available from http://www.sciencedirect.com
% rpb203 to LNT 20020327 (as rpb206)
% rpb203

\bibitem{rpb204} \wfl, \rpb\ and Hong Shen,	% formerly 205
Fully dynamic maintenance of $k$-connect\-ivity in parallel,
{\em IEEE Trans.\ on Parallel and Distributed Systems} 12 (2001), 846--864.
% Submitted to IEEE Parallel and Distributed Systems in 1995, paper #D95179,
% editor Pearl Wang,
% referees reports 19990407 and 20000301, revised by Weifa
% Accepted subject to minor revisions suggested by referee (now done),
% according to email from Weifa 20010221, appeared August 2001
% See Liang/IPPS96.*, IPPS96TR.*, rpb162.ps, rpb162.pdf
% Final version is Liang/tpds01_final.ps.gz = rpb204.ps.gz
% Available from IEEE, 00946661.pdf -> rpb204.pdf
% rpb204 to LNT 20020322 (as rpb205)
% rpb204

\bibitem{rpb205} \lty\ and \rpb, % [76] of Laurence's transfer report, May 2003
Parallel execution time analysis for least squares problems on distributed
memory architectures,
{\em International Journal of Computer Research} 10, 4 (2001), 517--530.
% Hardcopy arrived October 2002 - first that I saw of it ?
% Laurence sent pdf file ijcr-final-09.pdf 20021114
% See http://www.cs.okstate.edu/~ijcr/Vols/vol10no4.html
% rpb205 to LNT 20030321
Revision appeared as
``Parallel time complexity for solving large and sparse least squares problems
on distributed memory architectures'' in
{\em Practical Parallel Computing}, 
% edited by Marcin Paprzycki, Luciano Tarricone and Laurence Tianruo Yang,
Nova Science Publishers, New York, 2001, 97--112.
% Information from Laurence 20030127.  He says the two versions are
% almost the same. The second version is a book of selected papers
% (also called a special issue of the journal IJCR)
% rpb205

\bibitem{rpb206} \lty\ and \rpb, % [77] of Laurence's transfer report, May 2003
The improved BiCGStab method for large and sparse unsymmetric linear systems
on parallel distributed memory architectures,
{\em Proc.\ Fifth International Conference on
Algorithms and Architectures for Parallel Processing} (ICA3PP-02),
Beijing, October 2002, 324--328.
% Did Laurence go to Beijing to present the paper ?
% rpb206.pdf = ica3pp02-final02.pdf from Laurence 20021119
% Available from IEEE, 01173595.pdf -> rpb206i.pdf
% Problem printing rpb206*.pdf with Acroread 4 (on gauss)
% but OK with Acroread 3 (on portland).
% rpb206 to LNT 20030321.
% Following is [85] of Laurence's transfer report, May 2003
% Revision ``The parallel IBiCGStab method for linear systems 
% on parallel distributed memory architectures'', to appear in 
% {\em Journal of Parallel Processing Algorithms and Applications}, 2004.

\bibitem{rpb207} \bbz\ and \rpb,
An efficient method for computing eigenvalues of a real normal matrix,
\JPDC\ 63 (2003), 638--648.
% Accepted 20021111
% Available electronically 20030620
% rpb207 to LNT 20030321.
% Based on revision of rpb163
% rpb207

\bibitem{rpb208} \lty\ and \rpb, % [81] of Laurence's transfer report, May 2003
The improved parallel 			% "parallel" in title of ps version
{BiCG} method for large and sparse linear systems on parallel
distributed memory architectures,
{\em Information Journal} (special issue on parallel computing),
6, 3 (2003), 349--360.
MR~2004h:65037.
% ipdps.pdf, draft of 20020123
% (new version of IPDPS-01 ?)
% Corrections to Laurence 20020327 - did they get in final version ?
% paper accepted for special issue of Information Journal
% (page charges $US360 - contribute from EPSRC grant ?)
Preliminary version appeared 
%as
%% [78] of Laurence's transfer report, May 2003
%``The improved biconjugate gradient method on parallel distributed memory
%architectures'', 
in
{\em Workshop Proceedings of the 16th International Parallel
and Distributed Processing Symposium} (IPDPS-PDSECA02),
Fort Lauderdale, Florida, April 2002, 233--240.	% 15--19 April 2002
% Available from IEEE (fee), no "parallel" in title, page 233b,
% see Yang/IPDPS-02.pdf
% Reference [15] of rpb206
% See rpb208.pdf (conference version)
% rpb208[.pdf] to LNT 20030321.
% rpb208.ps from Laurence 20030331 (journal version)
% rpb208

\bibitem{rpb209} \bbz, A.~M.~Goscinski and \rpb,
Concerning the length of time slots for efficient gang scheduling,
{\em Proc.\ Third International Conference on Parallel and Distributed
Computing, Applications and Technologies} (PDCAT2002),
Kanazawa, Japan, September 2002, 364--371. 
% IEEE CS Press
% ISBN 4-9900330-2-7.
% Bing presented at conference in Japan, Sept 2002, see email of 20021112.
% Files at Bing/PDCAT02/* and 
% Bing/ZhouInvitedIEICE.pdf (appeared in IEICE Transactions).
% rpb209 to LNT 20030321.
Revision in {\em IEICE Transactions on Information and Systems}
{E86-D} (2003), 1594--1600. % Vol.E86-D No.9 pp.1594-1600 2003/9
% See http://www3.oup.co.uk/ietisy/ and http://search.ieice.or.jp/index-e.html
% pdf available from
% http://search.ieice.or.jp/2003/files/e000d09.htm#e86-d,9,1594
% rpb209

\bibitem{rpb210} \lty\ and \rpb, % [75] of Laurence's transfer report, May 2003
The improved conjugate gradient squared (ICGS) method on parallel
distributed memory architectures, in {\em Workshop Proceedings of the 2001
International Conference on Parallel Processing} (ICPP-HPSECA01),
Valencia, Spain, Sept.\ 2001, 161--165.
% Reference [14] of rpb206
% See ICPP-01 (missing), draft of 20020118, is this final ?
% Available from IEEE, 00951924.pdf -> rpb210i.pdf
% rpb210 to LNT 20030321.
% Revision ``The improved ICGS method for large and sparse unsymmetric linear
% systems on distributed memory architectures'' to appear in 
% {\em Journal of Computational Methods in Sciences and Engineering}.
% See http://www.cisp-publishing.com/jcmse.html
% Information from Laurence 20030127, but did not appear in 2003.
% Compare similar-sounding [84] of Laurence's transfer report, May 2003,
% The improved parallel ICGS method for large and sparse unsymmetric
% linear systems on distributed memory architectures,
% to appear in Journal of Parallel Processing Letters, 2004 (?)

\bibitem{rpb211} \rpb\ and \pz,
Random number generators with period divisible by a Mersenne prime
(keynote address),
in {\em Computational Science and its Applications~-- ICCSA 2003}
%{\em Proc.\ International Conference on Computational Science
%and its Applications}
%(ICCSA~2003, Montreal, May 2003),
% edited by V.~Kumar \etal,
%% Vipin Kumar, Marina L. Gavrilova, Chih Jeng Kenneth Tan, Pierre L'Ecuyer,
\LNCS, Vol.~2667, Springer-Verlag, Berlin, 2003, 1--10.
ISBN 3-540-40155-5.
% MR2110688.
% Invited/keynote paper, ICCSA Montreal 2003
% Draft to PZ 20030126,
% final version Montreal/CFBRENT.tar.gz submitted 20030311.
% Montreal/rpb211.tex
% rpb211 to LNT 20030321.

\bibitem{rpb212} \rpb\ and \pz,
Algorithms for finding almost irreducible and almost primitive trinomials,
in {\em High Primes and Misdemeanours:
Lectures in Honour of the $60$th Birthday of Hugh Cowie Williams},
edited by A.~van der Poorten and A.~Stein,
Fields Institute Communication FIC/41,
The Fields Institute, Toronto, 
% published by the American Mathematical Society,
2004, 91--102.
% ISBN 0-8218-3353-7, ISSN: 1069-5265
% DOI: http://dx.doi.org/10.1090/fic/041
MR~2005f:11291.	% MR2070919
Also arXiv:2105.06013.	% 13 May 2021 (but version dated 2003)
% arXiv pwd zvsgd
% For Banff conference, 60th birthday of Hugh Cowie Williams, 24-30 May 2003.
% Final version (5.13) to Alf 20030417, corrections sent 20030823
% See http://www.ams.org/cgi-bin/bookstore/bookpromo/ficseries
% and http://www.ams.org/bookstore?fn=20&arg1=ficseries&item=FIC-41
% Expected publication date is 20040619.
% Banff/rpb212.tex
% rpb212 to LNT 20030321.
% Also \url{https://hal.inria.fr/inria-00099724}, 20060926 (abstract only)
% Submitted the ``preprint'' version to arXiv; it appeared with date 
% 13 May 2021 but copyright 2003 as in the original.  
% Later add publication info. (done 20210514).

\bibitem{rpb213} \rpb,
Some comments on C.~S.~Wallace's random number generators,
{\em {Computer Journal}} 
51, 5 (C.~S.~Wallace memorial special issue, Sept. 2008), 579--584.
arXiv:1005.2314,
% Published electronically 15 February 2008.         
% Available from
% http://comjnl.oxfordjournals.org/papbyrecent.dtl 
\url{http://doi.org/10.1093/comjnl/bxm122}.
% see Wallace/rpb213.tex
% Sent (version 1.00) to David Dowe 20030321.
% Accepted (with minor changes to be sent by 20030614) 20030512
% rpb213 version 1.00 to LNT 20030321.
% Revised 20030514, sent "final" version to Dowe 20030613.
% See David Dowe's email of 20071010.
% Sent slightly revised final version to Dowe 20071017.
% Sent copyright form 20080113.
% See Dowe's email of 20080801 re cover of special issue.
% rpb213

\bibitem{rpb214} \rpb, S.~Larvala and \pz,
A primitive trinomial of degree 6972593,
\MC\ 74 (2005), 1001--1002. %74, 250 (April 2005) 
MR2114660 (2005h:11054)
% Posted 25 May 2004.
% Submitted to Math Comp 20030826, ref P-9495, 
% see random/mathcomp/BLZ6note/BLZ6note.tex
% Accepted subject to revisions 20030922
% Revised version BLZ6note-rev3 sent to Math Comp 20031006, accepted 20031007
% Final version random/mathcomp/BLZ6note/P-9495.tex to AMS via ftp 20031028
% Officially received 20031106.
% AMS ID BRRCPK, article ID MCOM1673, see www.ams.org/mstrack/
% Returned proofs to AMS 20040513 (no corrections)
% Appeared on AMS web site 20050114, 
% ref S0025-5718-04-01673-4BRRCPK, see http://www.ams.org/mcom
% rpb214
% Extended version to appear as
% Research Report, INRIA Lorraine, Villers-l\`es-Nancy, France.							
% Has not appeared yet as waiting to include details on checking and <<<===
% fast gcd							     <<<===
% Did it ever appear as an INRIA report ? Maybe not!		     <<<===
% rpb214tr

\bibitem{rpb215} \lty\ and \rpb, 
% Following apparently presented by Laurence (email 20030127)
% [80] of Laurence's transfer report, May 2003
% Title actually "The improved ..." but omit "The"
Improved Krylov subspace methods for large and sparse linear systems
on Bulk Synchronous Parallel architectures,
{\em \Proc\ 17th International Parallel and Distributed Processing Symposium}
(IPDPS03), IEEE Computer Society, NY, 2003, 11\pp
% {\tt http://csdl.computer.org/comp/proceedings/ipdps/2003/}~. 
%
% Workshop Proceedings of the 17th International Parallel and Distributed
% Processing Symposium (IPDPS-PDSECA03), Nice, France, 22--26 April 2003.
% (PDSECA = Workshop 1G - Parallel and Distributed Scientific and
%  Engineering Computing with Applications)
% According to http://www.informatik.uni-trier.de/ and http://
% www.informatik.uni-trier.de/~ley/db/conf/ipps/ipdps2003.html#YangB03
% this appeared in IPDPS 2003, 260b (this is the paper #)
% Available from http://
% csdl.computer.org/comp/proceedings/ipdps/2003/1926/00/19260260babs.htm
% Downloaded pdf - see IPDPS-03.pdf (243522 bytes)
% I did not see this before publication!  Don't make pdf available.

\bibitem{rpb216} \lty\ and \rpb,
% Following is reference [83] of Laurence's transfer report, May 2003,
% Who presented the paper ?
Parallel MCGLS and ICGLS methods for least squares problems on
distributed memory architectures,
{\em Proceedings of the 2003 International Symposium on Parallel and 
Distributed Processing and Applications} (ISPA~2003), 
% Aizu-Wakamatsu City, Japan, 2--4 July 2003,  
% M. Guo and L. T. Yang (editors),
\LNCS, Vol.~2745, Springer-Verlag, Berlin, 2003, 197--208.
% MR2050340.
% Available from Springer http://www.springerlink.com/
% Downloaded file IPSA-03.pdf (198231 bytes)
%
% Paper with the same title so presumably almost the same:
Revision in
{\em The Journal of Supercomputing} 29 (2004), 145--156. 
% 29(2), August 2004, 145--156,
% see 
\url{http://dx.doi.org/10.1023/B:SUPE.0000026847.75355.69}.
% and email from Laurence 20040512.
% Special Issue: Parallel and Distributed Processing with Applications
% (Guest Editor: Minyi Guo).
%
% I did not see these before publication!  Don't make pdf available.

\bibitem{rpb217} \rpb,
Fast and reliable random number generators for scientific computing,
{\em Proceedings of the PARA'04 Workshop on the
State-of-the-Art in Scientific Computing} (Lyngby, Denmark, June 2004),
\LNCS\ Vol.\ 3732, Springer-Verlag, Berlin, 2006, 1--10
(invited talk). 
% Appeared electronically 20060301, see rpb217-LNCS.pdf
Extended abstract in the Preliminary Proceedings (CDROM), Volume 1, 3--9.
% See Lyngby/PARA04-7pg.* for extended abstract, as on CDROM, V.1, 3-9.
% rpb217a
% see Lyngby/PARA04.* for full paper, emailed 20040914, appeared in the
% final Proceedings.  Final corrections to Jerzy 20051028.
MR2264001 (2008e:65011)
% rpb217

\bibitem{rpb218} \rpb,
Note on Marsaglia's xorshift random number generators,
{\em Journal of Statistical Software} 11, 5 (2004), 1--4.
% Submitted 20040707, see email to de Leeuw, mail/JSS,
% Acknowledged 20040707, ref JSS 138
% Accepted and appeared 20040822, see email from de Leeuw
% random/xorshift-note.tex

\bibitem{rpb219} \rpb,
Brain drain: a more positive note,
{\em Australian Math.\ Soc.\ Gazette} 31, 4 (Sept.\ 2004), 234--238.
% http://www.austms.org.au/Gazette
% The fourth (and last) in a series of essays on the ``Brain Drain''.

\bibitem{rpb220} \bbz, M.\ Tarawneh, C.\ Wang, A.\ Zomaya and \rpb,
% Bing Bing Zhou, Monther Tarawneh, Chen Wang, Albert Zomaya, rpb
A novel quartet-based method for phylogenetic inference,
{\em Proc.\ Fifth IEEE Symposium on Bioinformatics and Bioengineering} 
(BIBE05), % University of Minnesota, 19-21 Oct 2005
IEEE CS Press, New York, 2005, 32--39.  % 19-21 Oct 2005
% See email from Bing 20050731. 
% Presented by Bing at Minneapolis, 20051020 ?
% Proceedings to be published by IEEE CS Press.
% Also poster at IEEE Computational Systems Bioinformatics Conference
% (CSB2005), California, Aug. 2005 (according to Bing's report on
% ARC Project DP0557909).
% Bing/BIBE05P39.pdf (converted from .doc)
% Appeared, pp 32--39.

\bibitem{rpb221} \rpb, Colin Percival and \pz, 
Error bounds on complex floating-point multiplication,
\MC\ 76 (2007), 1469--1481.
MR2299783 (2008b:65062).
% Submitted to Math Comp 20051121 (by Colin Percival), ref P-10171.
% Finally accepted 20060412 (see emails), posted 20070124.
% Percival/complexmultiply-MC.tex
Also (extended version) Report RR-6068, INRIA, France,
\url{http://hal.inria.fr/inria-00120352/en/}, Dec.~2006, 25~pp.
% Also INRIA Technical Report #6068, published about 20061220,
% see rpb221tr.pdf, http://hal.inria.fr/inria-00120352/en/

\bibitem{rpb222} \bbz, D.\ Chu, M.\ Tarawneh, P.\ Wang, C.\ Wang,
A.\ Y.\ Zomaya and \rpb,
Parallel implementation of a quartet-based algorithm for phylogenetic analysis,
{\em Proceedings Fifth IEEE International Workshop on High Performance
Computational Biology} (HiCOMB06), Rhodes, Greece, April 2006, 8~pp.
\url{http://www.hicomb.org/papers/HICOMB2006-04.pdf}
% No page numbers as on CD, but 8 pp.
% 20th IPDPS 2006. 25-29 April 2006 -
% Digital Object Identifier   10.1109/IPDPS.2006.1639534
% To appear in {\em IEEE Bioinf.} % ? Did this appear ?
% {\em IEEE/ACM Trans.\ on Computational Biology and Bioinformatics}.
% In conjunction with IPDPS06.
% Accepted 20051220. See email from Bing 20051221. 
% Bing/HiCOMB06.pdf (not yet final version).
% Final version was due 20060121.
% According to google, this appeared online in the Proceedings pp 1-8
% and is to appear in IEEE Bioinf.
% See Bing's email of 20061106 and file HiCOMB06paper.zip (to be retrieved)

\bibitem{rpb223} \bbz, M.\ Tarawneh, D.\ Chu, P.\ Wang, C.\ Wang,
A.\ Y.\ Zomaya and \rpb,
On a new quartet-based phylogeny reconstruction algorithm,
{\em Proceedings
2006 International Conference on Bioinformatics and Computational Biology}
(BIOCOMP06), Las Vegas, Nevada, USA, June 2006, 87--93.
%% Paper BIC4097, conference held at Las Vegas, 26-29 June 2006,
%% 2006 International Conference on Bioinformatics and Computational Biology
%% see email of 20060402.
% See Bing's email of 20061106 and biocomp06-BIC4097.pdf -> rpb223.pdf 

\bibitem{rpb224} \rpb, 
Some long-period random number generators using shifts and xors,
% presented at
% 13th Biennial Computational Techniques and Applications Conference
% (CTAC06), Townsville, 3 July 2006.
{\em ANZIAM J.} 48 (CTAC2006), C188--C202, 2007.
% Presented at CTAC06 20060703,
% submitted online to ANZIAM J 20060706.
% Accepted 20070201, see email from Wayne Read
% See random/CTAC06
% Uploaded revised version brent.tar = 40-764-3-CE.tar 20070702.
% Published electronically 20070706
% \url{http://anziamj.austms.org.au/ojs/index.php/ANZIAMJ/article/view/40},
arXiv:1004.3115 % pwd kw57w

\bibitem{rpb225} \rpb,
Fast algorithms for high-precision computation of elementary functions
(invited talk: extended abstract), 
Proc.\ Seventh Conference on Real Numbers and Computers (RNC7), Nancy,
France, 10-12 July 2006, 7-8.
% See sources/RNC7-Brent.tex for extended abstract, slides in talks/...
% Proceedings at http://rnc7.loria.fr/rnc7-proceedings.pdf

\bibitem{rpb226} \rpb\ and \pz, {\em Modern Computer Arithmetic},
Cambridge Monographs on Applied and Computational Mathematics (No.~18),
Cambridge University Press, 2010, 236 pages. % published Nov 2010
% ISBN 9780521194693
MR2760886 (2012h:65315), 
% Reviewed by Song Yan in ACM-SIGACT News, No 4, Vol 43, 2012;
% also in SIAM Review 53, 4 (December 2011), 809-810 by Warren E. Ferguson.
% Submitted version 0.5.1 (5 March 2010) to arXiv 20100427
\url{http://doi.org/10.1017/cbo9780511921698.001},
arXiv:1004.4710. % pwd zcdzs

\bibitem{rpb227}
\bbz, M. Tarawneh, D. Chu, P. Wang, C. Wang, A. Zomaya and \rpb,
Evidence of multiple maximum likelihood points for a phylogenetic tree,
{\em Proceedings Sixth IEEE Symposium on BioInformatics and BioEngineering}
(BIBE2006),
Arlington, Virginia, USA, Oct.~2006, 193--197. 
% See emails from Bing 20061026 and 20061106
% and BIBE06-P9.pdf -> rpb227.pdf

\bibitem{rpb228}
P. Wang, \bbz, M. Tarawneh, D. Chu, C. Wang, A. Zomaya and \rpb,
A global maximum likelihood super-quartet phylogeny method,
{\em Proceedings Fifth Asia-Pacific BioInformatics Conference} (APBC2007),
Hong Kong, Jan.~2007, 111--120.
% Pinghao Wang, Bing Bing Zhou, Monther Tarawneh, Daniel Chu, Chen Wang,
% Albert Y. Zomaya, Richard P. Brent: A Global Maximum Likelihood
% Super-Quartet Phylogeny Method. APBC 2007: 111-120 
% See email from Bing 20061026

\bibitem{rpb229}
Y. J. Lui, \rpb\ and A. Calinescu, % Yuan J. Lui, Richard Brent & Ani Calinescu
% Note that order is Lui, Calinescu, Brent in the IEEE version
Extracting significant phrases from text,
% {\em Proceedings 2007 IEEE International Symposium on Data Mining and
% Information Retrieval} (IEEE DMIR-07), % Niagara Falls, May 21-23, 2007
% Ontario, Canada, May 2007, to appear.
% to be published by IEEE CS Press.
% see Lui's email of 20070210
% and http://www.cs.wisc.edu/dbworld/messages/2006-08/1157046398.html
% Also 21st International Conference on Advanced Information Networking and
% Applications (AINA 2007), Workshops Proceedings, Volume 1, May 21-23, 2007,
% Niagara Falls, Canada. IEEE Computer Society 2007
{\em Proc.\ 21st International Conference on Advanced Information 
Networking and Applications} (AINA 2007), 
{\em Workshop Proceedings} (AINAW07),
Vol.~1, May 2007, 361--366.
\url{http://doi.ieeecomputersociety.org/10.1109/AINAW.2007.180}.

\bibitem{rpb230}
\rpb\ and \pz, % Richard P. Brent & Paul Zimmermann
% "distinct degree" in INRIA report but "distinct-degree" in Fq8 Proc.
A multi-level blocking distinct-degree factorization algorithm,
in {\em Finite Fields and Applications} (edited by G.~Mullen, D.~Panario,
and I.~Shparlinski),
{\em Contemporary Mathematics}, Vol.~461, 2008, 47--58.
arXiv:0710.4410.
MR2436324 (2009k:11041)
% presented at the {\em Eighth International Conference on Finite Fields and
% Applications}, % Fq8, 
% Melbourne, 9-13 July 2007. 
% http://www.deakin.edu.au/scitech/eit/fq8/
% Proceedings to be published after the conference, deadline 20071012.
% Sent pdf file to Gary Mullen 20071010, see Fq8/Mullen/Fq8-paper.pdf
% Fq8/Fq8-paper.tex; for abstract see slides/Fq8-abstract.tex
% To appear in the Proceedings, {\em Contemporary Mathematics}, special issue.
% Finite Fields and Applications, 
% Edited by: Gary L. Mullen, Daniel Panario, and Igor E. Shparlinski, 
% Contemporary Mathematics 2008; 265 pp; softcover Volume: 461
% ISBN-10: 0-8218-4309-5, ISBN-13: 978-0-8218-4309-3
%
% Also appeared as INRIA Tech Report RR-6331,
% \url{https://hal.inria.fr/inria-00181029/en}, October 2007, 16~pp;
% and at arXiv:0710.4410.
% \url{http://arxiv.org/abs/0710.4410}~.
% (see emails from PZ 20071009, 20071024).

\bibitem{rpb231}
Shi Bai and \rpb,
On the efficiency of Pollard's rho method for discrete logarithms,
{\em The Australasian Theory Symposium} (CATS2008),
Wollongong, 22--25 Jan.~2008. {\em Conferences in Research and
Practice in Information Technology}, Vol.~77, edited by James Harland and
Prabhu Manyem, Australian Computer Society, 2008, 125-131.
% Published on CDROM.

\bibitem{rpb232}
\rpb, Pierrick Gaudry, Emmanuel Thom\'e and \pz,
Faster Multiplication in ${\mathrm GF}(2)[x]$, 
{\em ANTS VIII 2008}, edited by A.~J.~van der Poorten and A.~Stein,
\LNCS\ 5011, 153--166.
MR2467844 (2009j:11197).
% submitted to (and accepted for)
% ANTS 8 paper, Banff, 17-22 May 2008,
% length max 14 pages in LNCS style, deadline for submission 13 Nov 2007.
% submitted by Emmanuel, ID 132, pwd 7JDoPdem-K
% Presented by Pierrick Gaudry at ANTS 8, Banff, May 2008.
% svn/mulgf2/mulgf2.*
Also INRIA Tech Report RR-6359,
\url{http://hal.inria.fr/inria-00188261/en/}, November 2007, 19~pp.
% paperId: inria-00188261, pwd: y7ppjfe

\bibitem{rpb233}
\rpb\ and \pz, % Richard P. Brent & Paul Zimmermann
Ten new primitive binary trinomials,
\MC\ 78 (2009), 1197--1199. 
MR2476580 (2010a:11040).
% Accepted 16 April 2008, posted electronically 1 August 2008:
% \url{http://www.ams.org/mcom/0000-000-00/S0025-5718-08-02170-4/}
% Reference P-10978
% Submitted 20080415, accepted 20080416, sent Latex file 20080416
% Page proofs received 20080531
% See book/irred/32582657.tex and sources/rpb233.tex

\bibitem{rpb234}
\wfl, \rpb, Yinlong Xu and Qingshan Wang,
Minimum-energy all-to-all multicasting in wireless {\em ad hoc} networks,
% submitted to IEEE TWC.
% Revision submitted about 20071105, see emails from Weifa 20071103,
% Second revision submitted about 20080724, ref Paper-TW-Jun-07-0602.R2,
% see email from Vijay Bhargava 20080724.
% ps/Liang/twc07.pdf
% Minor revisions requested 20090107, see email from Weifa
% Accepted 20090804, see email from Weifa, ref Paper-TW-Jun-07-0602.R3.
{\em IEEE Transactions on Wireless Communications}
8 (2009), 5490--5499. % 8, 11 (2009), 5490-5499.
% Proofs returned 20090825.
% See errata in email from Weifa (should be fixed in published version).

\bibitem{rpb235}
\rpb\ and \pz,
The great trinomial hunt,
{\em Notices of the American Mathematical Society} 58, 2 (2011), 233--239.
Also {\em Mathematical Advances in Translation} (Chinese)
33, 3 (2014), 248--257.
% \#2 (February 2011)
% Appeared electronically by 20110113
% submitted 23 September 2009, accepted (see email from Krantz 20100104).
% See trinom/AMSNotices.tex, book/trinom/AMSNotices.tex,
% trinom/Brent-Zimmermann.tex
% submitted version without figure 1 to arXiv, 20100511
% Final version 
% https://www.ams.org/journals/notices/201102/rtx110200233p.pdf
MR2768116. % no review
Also arXiv:1005.1967. % pwd vmez8
% Stephen Krantz forwarded to AMS 20101104 (see email)
% Returned proofs with corrections and copyright form to Sandy Frost 20101203
% 20120903: gave permission for a translation to appear in 
% "Mathematical Advances in Translation" sponsored by the Academy of
% Mathematics and Systems Science(Chinese Academy of Sciences).
% See email from Ms Huijuan Wang <whj@amss.ac.cn>
% See also email 20141103 from <liyanfang@math.ac.cn>
% Copy of Chinese translation arrived Nov. 2014.

\bibitem{rpb236}
\rpb\ and \pz,
An $O(M(n)\log n)$ algorithm for the Jacobi symbol,
Proceedings of the Ninth Algorithmic Number Theory Symposium (ANTS-IX)
held in Nancy, France, 19--23 July 2010.
\LNCS\ 6197, 2010, 83--95.
% accepted for {\em ANTS IX}, Nancy, July 2010.
MR2721414 (2011m:11248), % Review by Jeffrey O. Shallit
arXiv:1004.2091v2
% Submitted to ANTS9, 20100117. Accepted, see emails 20100401.
% Following info from PZ:
%@InProceedings{BrZi10a,
% author =       {Richard P. Brent and Paul Zimmermann},
% title =        {An {$O(M(n) \log n)$} algorithm for the {J}acobi symbol},
% booktitle = {Proceedings of the 9th Algorithmic Number Theory Symposium
%              (ANTS-IX)},
% address = {Nancy, France, July 19-23, 2010},
% year =         2010,
% series = lncs,
% volume = {XXXX},
% publisher = sv,
% editors = {Guillaume Hanrot and {Fran\c cois} Morain and Emmanuel
%{Thom\'e}},
% annote = {http://arxiv.org/abs/1004.2091}
%}
% Also submitted to arXiv 20100413, see Zimmermann/jacobi_arXiv.*
% arXiv:submit/0022061, arXiv:1004.2091v1, paper pwd g7kmd
% arXiv logon info: Richard.Brent@anu.edu.au, RichardPBrent, r..arxiv

\bibitem{rpb237}
\rpb, 
Note on computing ratings from eigenvectors, 5 May 2010, 10 pp.
arXiv:1005.0762v1

\bibitem{rpb238}
\rpb,
George Forsythe's last paper, 6 May 2010, 10 pp.
arXiv:1005.0909v2 % v2 corrected spelling of "Forsythe" in metadata

\bibitem{rpb239}
\rpb,
A simple approach to error reconciliation in quantum key distribution,
7 May 2010, 19 pp.
arXiv:1005.1206v1

\bibitem{rpb240}
\rpb,
The myth of equidistribution for high-dimensional simulation,
8 May 2010, 8 pp.
arXiv:1005.1320v1

% \bigskip\leftline{\bf Accepted, submitted or in preparation}

\bibitem{rpb241}
N. Nandapalan, \rpb, L. M. Murray and A. Rendell,
High-performance pseudo-random number generation on graphics processing units,
in Parallel Processing and Applied Mathematics
(Proc.\ PPAM 2011, Torun, Poland, 11--14 Sept.\ 2011), 
\LNCS, Vol.~7203 (2012), 609--618.
arXiv:1108.0486v1. % Submitted to arXiv 20110802, pwd xbvng
% http://ppam.pl 
% Submitted to PPAM2011, see emails.
% Accepted 20110703 (subject to changes suggested by referees), see emails
% and referees' reports at ps/Nandapalan/PPAM-referees.txt
% Source is at svn/PPAM2011-xorgensGP/*
% Submitted to arXiv after revisions made, before signing copyright agreement..
% http://www.springer.com/computer/swe/book/978-3-642-31463-6

\bibitem{rpb242}
\rpb\ and D.\ Harvey,
Fast computation of Bernoulli, Tangent and Secant numbers,
{\em Proceedings of a Workshop on Computational and Analytical Mathematics
in honour of Jonathan Borwein's \hbox{$60$-th} birthday,}
D.~H.~Bailey et al. (eds.),
{\em Springer Proceedings in Mathematics \& Statistics}, Vol.~50, 2013,
Chapter~8, 127--142.
% Submitted to Heinz Bauschke 20110801, see Borwein/jonfest/tangent.*
% Based on talk by rpb at Jon's 60th birthday workshop, May 2011.
% Accepted 20110824, subject to some additions suggested by the referees.
% Finally accepted 20110831, send latex file Brent-Harvey.tex to Bauschke.
% See sources/rpb242.tex, Borwein/jonfest/tangent.tex, Brent-Harvey.tex etc.
% Received page proofs 20130624, see email from NancyLydia,
% returned by 20130627, see Borwein/jonfest/218128_1_En_8_corrected.pdf
% Received the book 20131215
% DOI: 10.1007/978-1-4614-7621-4\_8, Springer, New York, 2013.
% http://link.springer.com/book/10.1007/978-1-4614-7621-4
MR3108426,
arXiv:1108.0286v3 % Submitted to arXiv 20110801, pwd itku2

\bibitem{rpb243}
L-T.\ Wang, N.\ A.\ Touba, \rpb, H.\ Wang and H.\ Xu,
% Laung-Terng Wang, Nur A. Touba, rpb, Hui Wang, Hui Xu,
{\em High-speed Hybrid Ring Generator Design Providing Maximum-Length
Sequences with Low Hardware Cost},
Tech.\ Report UT-CERC-12-01,
Computer Engineering Research Center,
The University of Texas at Austin, USA,
4 Oct 2011, 12\pp
% Submitted to 30th IEEE VLSI Test Symposium, see emails from LT Wang
% and Claude Thibeault 20111005. 
% Paper ID 139-BZ654, username rpbrent, pwd i..dac, website http://www.dac.com/
% Rejected (email from Thibeault 20111228).
% Similar paper "Minimum-Cost m-Sequence Linear Feedback Shift Register
% Design", paper ID 139-BZ654, was rejected by 49th DAC (see email 20120221).

\bibitem{rpb244}
\rpb, W.\ Orrick, J.\ Osborn, and P.\ Zimmermann,
Maximal determinants and saturated D-optimal designs of orders $19$ and
$37$, 
% Will Orrick submitted to Designs, Codes and Cryptography 20111219,
% publisher is Springer,  manuscript #DESI-1887, see emails.
% Submitted n19v21 (rpb244) to arXiv 20111218, see arXiv submit/0382925,
% also submitted rpb244_data.tgz containing some data files and programs,
% see Hadamard/papers/arxiv/* (arXiv pwd for this paper is 7p5pe)
% Reply from editor <janani.kalidasan@springer.com> dated 20120629,
% with three referees' reports.  Generally they are unhappy with our
% presentation of the results and ask for a major revision.
% As at 20190815, this has not been done.
arXiv:1112.4160v1, 18 Dec.\ 2011, 28 pp.
% Note added 20170311: it looks as if nothing is going to happen
% unless I (RPB) draft the revision.  Maybe simplify the description
% of the algorithm and omit the order $37$ case?  Also, could try
% a different journal such as AJC or Maple Transactions or JIS.
% 20210815: Still unable to contact Will Orrick despite several attempts,
% including via Bruce Solomon and via Will's wife.

\bibitem{rpb245}
\rpb,
Finding D-optimal designs by randomised decomposition and switching,
{\em Australasian Journal of Combinatorics}, 
55 (2013), 15--30.
Erratum \url{http://maths-people.anu.edu.au/~brent/pub/pub245_errata.html}.
% For Australasian J. Combinatorics special issue on Hadamard matrices
% in honour of Kathy Horadam's 60th birthday. Submitted 20111220.
% Acknowledged, see email from Elizabeth Billington 20111221, ref. P410
% See Hadamard/papers/maxdet-RMIT.*, Brent-P410*.*, /rpb245/rpb245*.tex
% Corrected error in Table 5 20111222, see version 2.
% Updated Table 5 and improved analysis of it, see version 3.
% Corrected Figures, see version 4 (also Brent-P410-v2 to AJC).
%arXiv:1112.4671v1 %(20111220)
%arXiv:1112.4671v2  %(20111222)
%arXiv:1112.4671v3  %(20120102)
%arXiv:1112.4671v4  %(20120112)
% Received two referees' reports 20120731 - one favourable and the other
% not.  Despite this, it seems from the editor's email that the paper is
% accepted subject to minor revisions.. 
% See Hadamard/papers/P410*, Hadmard/papers/rpb245/*
% Sent v5 to AJC 20120811
% email from Elizabeth Billington 20120825 confirmed that the paper has
% been accepted.
% Proofs received 20121222, seem OK, reply soon.
% Tentative details: volume 55 (2013), 15-30.
% Appeared 20130203
% http://ajc.maths.uq.edu.au/pdf/55/ajc_v55_p015.pdf
MR3051516,
arXiv:1112.4671v5 %(20120811)

\bibitem{rpb246}
J. Arias de Reyna, \rpb\ and J.\ van de Lune, % Juan, RPB, Jan
A note on the real part of the Riemann zeta-function,
{\em Herman J.\ J.\ te Riele Liber Amicorum},
% J. A. J. van Vonderen (editor),
CWI, Amsterdam, Dec.\ 2011, 30--36.
% on the occasion of Herman te Riele's retirement from CWI, Jan.\ 2012.
% Sent to Susanne van Dam 20111012, see Riemann/AriasBrentLune.*
% Paper: http://oai.cwi.nl/oai/asset/19699/19699D.pdf
% Full text: http://oai.cwi.nl/oai/asset/19691/19691D.pdf
% See also http://repository.cwi.nl/search/fullrecord.php?publnr=19691
% arXiv:1112.4910v1		% Riemann/AriasBrentLune.*, 20111221
% arXiv pwd d563q
Errata: see arXiv:1112.4910v2	% Riemann/AriasBrentLune-corrected.*, 20140516

\bibitem{rpb247}
\rpb\ and J.\ van de Lune,
A note on P\'olya's observation concerning Liouville's function,
{\em Herman J.\ J.\ te Riele Liber Amicorum},
CWI, Amsterdam, Dec.\ 2011, 92--97.
% on the occasion of Herman te Riele's retirement from CWI, Jan.\ 2012.
% Sent to Susanne van Dam 20111005, see Riemann/BrentLunePolya.*
% Paper: http://oai.cwi.nl/oai/asset/19722/19722D.pdf
% Book: http://oai.cwi.nl/oai/asset/19691/19691D.pdf
% See also http://repository.cwi.nl/search/fullrecord.php?publnr=19691
arXiv:1112:4911v1

\bibitem{rpb248}
J. Arias de Reyna, \rpb\ and J.\ van de Lune, % Juan, RPB, Jan
On the sign of the real part of the Riemann zeta-function,
\emph{Number Theory and Related Fields (in memory of Alf van der Poorten)},
edited by J.~M.~Borwein, I.~Shparlinski and W.~Zudilin,
Springer Proceedings in Mathematics and Statistics Vol.~43,
Springer, New York, 2013, 75--97.
% ISBN 978-1-4614-6641-3
% http://link.springer.com/book/10.1007/978-1-4614-6642-0
%Proc.\ International Number Theory Conference in memory of
%Alf van der Poorten} (Newcastle, NSW, March 2012), (accepted 30 July 2012).
% Paper submitted to Proceedings 20120520, alf@rpbrent.com, r..alf
% Revision submitted after referee's report 20120727, accepted
% Copyright form returned 20121205. Corrected proofs returned 20130311
% Book arrived at MSI on or before 20130704.
% http://link.springer.com/book/10.1007/978-1-4614-6642-0
% arXiv:1205:4423v1 % 20120520, pwd f2u4w
MR3081035, % an informative review by Adam Harper
% DOI 10.1007/978-1-4614-6642-0 (the whole book)
% DOI 10.1007/978-1-4614-6642-0_3 (this paper)
arXiv:1205:4423v2 % 20120730, pwd f2u4w

\bibitem{rpb249}
\rpb\ and J.\ H.\ Osborn,
General lower bounds on maximal determinants of binary matrices,
\emph{The Electronic Journal of Combinatorics} {20(2)}, 2013, \#P15, 12~pp.
% Hadamard/minors/H_minors_v02, ejc@rpbrent.com, rpbrent, r..ejc
% arXiv:1208.1805v1 % 20120809, pwd gqxd2, H_minors_v02, to EJC
% arXiv:1208.1805v2 % 20120903, H_minors_v05
% arXiv:1208.1805v3 % 20120904, H_minors_v06
% 20121212: referee's report 2612-5073-1-RV from EJC, wants some changes
% 20121218: resubmitted H_minors_v07 with new Sec. 2 and Appendix.
% arXiv:1208.1805v4 % 20121218, H_minors_v07
% 20130404: referee's report 2612-6274-1-RV from EJC, wants more changes.
% 20130405: resubmitted H_minors_v09 with new Sec. 5 and other minor changes.
% Provisionally accepted 20130405, see email from Brualdi
% 20130414: submitted final version rpb249_ejc.tex, .pdf to EJC using
% e-jc.sty style file etc. Content is almost the same as for H_minors_v10.
% Officially submitted 20120809, accepted 20130414, published 20130424
% http://www.combinatorics.org/ojs/index.php/eljc/article/view/v20i2p15
% arXiv:1208.1805v5 % 20130405, H_minors_v09
MR3066354,
% DOI: https://doi.org/10.37236/2612
arXiv:1208.1805v6 % 20130414, H_minors_v10
% Hadamard/minors/H_minors_v10, Hadamardminors/rpb249_ejc 

\bibitem{rpb250}
\rpb\ and J.\ H.\ Osborn,
Bounds on minors of binary matrices,
\emph{Bull.\ Austral.\ Math.\ Soc.\ }{88} (2013), 280--285.
% received 20120819, accepted 20120913
% Published online 26 Nov.\ 2012 (or 23 Dec.\ 2012).
% rpb250-preliminary.pdf
% http://journals.cambridge.org/abstract_S000497271200086X
% Hadamard/minors/B_minors_v03.tex
% submitted 17 August 2012, accepted 13 Sept 2012.
% To Bull. Austral. Math. Soc., ID 16858, #5953,
%  username richardpbrent,
% pwd 5785810, see Hadamard/minors/rpb250.*
%
% Also arXiv:1208.3330v1 % 20120816, pwd ju83w
% arXiv:1208.3330v2 % 20120817
MR3109717.
Longer version: arXiv:1208.3330v3, 9~pp. % 20120917, includes Sec. 4

\bibitem{rpb251}
\rpb\ and J.\ H.\ Osborn,
On minors of maximal-determinant matrices, 
\emph{Journal of Integer Sequences} 16 (2013), Article 13.4.2, 30~pp.
% https://cs.uwaterloo.ca/journals/JIS/VOL16/Brent/brent2.html
% submitted 19 August 2012.
% Hadamard/minors/minors_v08.tex submitted to J. Integer Sequences
% Accepted subject to minor revisions, see email from Shallit 20130304
% Revision to Shallit 20130311 (Brent_JIS_rev1.tex = minors_v10.tex)
% appeared 20130311 (see emails)
% https://cs.uwaterloo.ca/journals/JIS/VOL16/Brent/brent2.html
%arXiv:1208.3819v1 % 20120819, pwd dy352, minors_v08
%arXiv:1208.3819v2 % 20120831, pwd dy352, minors_v09
MR3056626,
arXiv:1208.3819v3, 10 March 2013. % 20130310, pwd dy352, minors_v10

\bibitem{rpb252}
\rpb, Old and new algorithms for $\pi$,
\emph{Notices of the AMS} (letter to the editor) 60, 1 (Jan.\ 2013), 7.
% accepted 22 September 2012.
% See letters/notices_letter.*, sources/rpb252.tex, pd/rpb252.pdf
% http://www.ams.org/notices/201301/rnoti-p7.pdf
arXiv:1303.2762v1. % 12 March 2013.

\bibitem{rpb253}
\rpb, J.\ H.\ Osborn and W.\ D.\ Smith,
Lower bounds on maximal determinants of $\pm 1$ matrices
via the probabilistic method,
%arXiv:1211.3248v1, 14 Nov.\ 2012. % 20121114, pwd pmera 
%arXiv:1211.3248v2, 3 Dec.\ 2012. % 20121203, pwd pmera
arXiv:1211.3248v3, 5 May 2013. % 20120505
% See Hadamard/minors/D_bound9.*
% This is a preliminary version of rpb257

\bibitem{rpb254}
Shi Bai, \rpb\ and E.\ Thom\'e, % Emmanuel Thom\'e
Root optimization of polynomials in the number field sieve,
%to appear in \emph{Math.\ Comp.} (accepted 21 Dec 2013).
\MC\ 84 (2015), 2447--2457.
% Number 295, Sept. 2015
% (published electronically 11 Feb.\ 2015).
MR3356034. % Review by J. D. Dixon
Also arXiv:1212.1958v1, 10 Dec.\ 2012, 16~pp. % pwd f52t4
% See ps/Bai/rs.*, ~/svn/rs/*
% Also Cryptology ePrint Archive (http://eprint.iacr.org/).
% SUBMISSION NAME: xxxx/819
% TITLE: Root Optimization of Polynomials in the Number Field Sieve
% AUTHORS: Shi Bai and Richard P. Brent and Emmanuel Thom\'e
% KEYWORDS: factoring, number field sieve, polynomial selection
% CONTACT: shih.bai@gmail.com
% PASSWORD: BB92W0
% See http://eprint.iacr.org/2012/691
% Submitted to Math Comp 20130614, see svn/rs/rs_mcomp/rs_mpcomp.*
% MCOM 130614-Bai
% Referee's reprt received 20130901 - paper needs to be revised and
% resubmitted.
% Version 2 submitted 20131030 (but my suggested improvements did not
% make it into this version - add them later if there is an opportunity).
% Submitted version is 130614-Bai-v2.pdf, cover letter 130614-Bai-c2.pdf
% Accepted by Math. Comp. 20131221
% Signed copyright form (electronically), MCOM2926, see email 20131228.
% DOI: 10.1090/S0025-5718-2015-02926-3
% Also appeared as an ANR CATREL report, see
% http://catrel.loria.fr/publications.fr.html

\bibitem{rpb255}
\rpb\ and J.\ H.\ Osborn,
Note on a double binomial sum relevant to the Hadamard maximal determinant
problem,
% arXiv:1309.2795v1, 11 Sept. 2013, pwd 83ifh
arXiv:1309.2795v2, 12 Sept.\ 2013, 6~pp. % typo in v1 corrected in v2
% Hadamard/minors/binomial_sum2.tex

\bibitem{rpb256}
\rpb\ and F.\ Johansson,	% Fredrik Johansson
A bound for the error term in the Brent-McMillan algorithm,
\MC\ 84 (2015), 2351--2359.
MR3356029. % Review by AllanJ. MacLeod
% (accepted 5 Jan.\ 2014). Appeared in Sept. issue, no. 295
% (published electronically 4 March 2015).
% As at 20150223 it is listed under
% ``Articles accepted prior to AMS Early View implementation''
Also arXiv:1309.0039v1, 29 Nov.\ 2013, 10~pp.
% arXiv pwd smvhu
% See ps/Johansson/bound_v6.* (v6 is 20131130, but 20131129 US time)
% Math Comp submitted 20131129, id 71337, see ~/ps/Johansson/MathComp/*
% 131129-Brent-v1.pdf = Brent-Johansson.pdf
% Referee's report, see email 20131231
% For revision see Brent-Johansson_rev1, 131129-Brent-v2.pdf,
% submitted 20140101 (or 20131231, depending on time zone).
% Accepted 20140105 (see email from Susanne Brenner), paper id MCOM2931.
% Submitted ps/Johansson/MathComp/Brent-Johansson_final.tex 20140107
% Signed online copyright form 20140108.
% Returned proofs to Luann Cole 20141217.
% DOI: 10.1090/S0025-5718-2015-02931-7
% Erratum: the inequality (2.7) for $I_0(x)$ holds for $x \ge 1$, 
% but not for $x \ge 0$ as claimed. This does not affect anything
% subsequently. 
Erratum:
\url{http://maths-people.anu.edu.au/~brent/pub/pub256_errata.html}

\bibitem{rpb257}
\rpb, J.\ H.\ Osborn and W.\ D.\ Smith,
Lower bounds on maximal determinants of binary matrices
via the probabilistic method,
%arXiv:1402.6817v1, 27 Feb.\ 2014. % Hadamard/minors/D_bound10.*
%arXiv:1402.6817v2, 14 Mar.\ 2014. % Hadamard/minors/D_bound11.*
%arXiv:1402.6817v3, 26 Jan.\ 2015.  % Hadamard/minors/D_bound11.*
%arXiv:1402.6817v4, 1 Mar.\  2015.  % Hadamard/minors/D_bound11.*
%arXiv:1402.6817v5, 26 Jan.\  2016. % sources/arXiv/rpb257v5.tex
arXiv:1402.6817v6, 25 Oct.\  2016.  % sources/arXiv/rpb257v6.tex
% arXiv pwd t56vu
% For preliminary version see rpb253
% Submitted 20140314 to EJC (see Hadamard/minors/rpb257_ejc.*), ms 4188
% Username rpbrent, pwd r..ejc
% Sent email enquiry to Andre Kundgen 20140922
% Brualdi sent reminder to referee 20140923 (see emails)
% Rejected (arXiv v2 or D_bound11) - see email from Brualdi 20140930
For a shorter version see \#262 below.

\bibitem{rpb258}
\rpb, J.\ H.\ Osborn and W.\ D.\ Smith,
Note on best possible bounds for determinants of matrices close to the
identity matrix,
\emph{Linear Algebra and its Applications} 466 (2015), 21--26
% official date is 1 Feb 2015
(available online from 16 Oct.\ 2014). % (accepted 28 Sept.\ 2014).
% Submitted to LAA 20140320, see Hadamard/minors/LAA/LAA-S-14-00325.pdf
% ms ref no is LAA-D-14-00258
% To login use Firefox, http://ees.elsevier.com/laa/,
% email/username pert@rpbrent.com, pwd i..c.LAA1
% Sent email query 20140703 - still under review
% Submitted for publication.
% Logged onto LAA website 20140922 - appears that ms was rejected 20140905
% though I didn't notice email to this effect. (There was an email 20140906).
% Sent email to Brualdi 20140922 to ask if he wants a shorter version.
% Sent email to Warren, J-a 20140922 to ask their opinion.
% Email reply from Brualdi 20140925 - willing to consider a short note.
% 20140926: submitted revision (with changed title),
% see LAA-D-14-00258R1.pdf and rpb258-LAA3.tex in Hadamard/minors/LAA/
% 20140928 - accepted, see email from Brualdi, ref LAA-D-14-00258R1
% 20141005 - filled out copyright transfer form and offprint order form
% (did not order offprints).
% Reference LAA 12920, 
% PII: S0024-3795(14)00638-7, 
% DOI: 10.1016/j.laa.2014.09.041
% Electronic proofs 20141006, returned electronically 20141011.
% Available online from 20141016 - see www/pd/rpb258-LAA.pdf
MR3278236. % Review by Youngkwon Song
Also (longer version)
\emph{Bounds on determinants of perturbed diagonal matrices},
arXiv:1401.7084v7, 27 Mar.\ 2014. % Hadamard/minors/det_bounds10.*
%arXiv:1401.7084v1, 28 Jan.\ 2014. % Hadamard/minors/det_bounds4.*
%arXiv:1401.7084v2, 29 Jan.\ 2014. % Hadamard/minors/det_bounds5.*
%arXiv:1401.7084v3, 25 Feb.\ 2014. % Hadamard/minors/det_bounds6.*
%arXiv:1401.7084v4, 28 Feb.\ 2014. % Hadamard/minors/det_bounds7.*
%arXiv:1401.7084v5, 14 Mar.\ 2014. % Hadamard/minors/det_bounds8.*
%arXiv:1401.7084v6, 19 Mar.\ 2014. % Hadamard/minors/det_bounds9.*
% pwd t4mep, arXiv date 13 March 2014.

\bibitem{rpb259}
\rpb,
Generalising Tuenter's binomial sums,
% rpb259v1.tex, paper pwd p88bq, 20140714
% rpb259v2.tex, 20140715
% rpb259v3.tex, 20140716
% rpb259v4.tex, 20150119 (minor corrections)
% Hadamard/minors/centred_sums2.tex
% Submitted to JIS, see Hadamard/minors/Brent_JIS_v0.tex
% and email to Shallit 20140717.
% Acknowledged by Shallit 20140719, see his email for list of
% small Latex points to change.
% Referee's report received 20141129 - seems to be accepted subject
% to revisions suggested by referee.
% Submitted revision 20150119, see Hadamard/minors/Brent_JIS_v1.tex
% Submitted revision 20150125, see Hadamard/minors/Brent_JIS_v2.tex
\emph{Journal of Integer Sequences} 18 (2015), article 15.3.2, 9 pp.
% Appeared electronically 20150126
MR3326794. % Review by Roberto B Corcino 
Also (longer version) arXiv:1407.3533v4, 19 Jan.\ 2015, 17 pp.

\bibitem{rpb260}
\rpb,  
H. Ohtsuka, % Hideyuki Ohtsuka (spelt this way in another publication)
J. H. Osborn, % Judy-anne Heather Osborn
and H. Prodinger, % Helmut Prodinger,
Some binomial sums involving absolute values,
\emph{Journal of Integer Sequences} 19 (2016), article 16.3.7, 14 pp.
MR3486753, % Review by Y. H. Harris Kwong
%arXiv:1411.1477v1, 6 Nov.\ 2014. % pwd ywiqz, ~/sources/arXiv/rpb260v1.tex
arXiv:1411.1477v2, 31 Jan.\ 2016. % pwd ywiqz, ~/sources/arXiv/rpb260v2.tex
% Submitted for publication.
% See Hadamard/minors/FQ/BOOP-FQ-v1.tex,  Hadamard/minors/BOOP_v3.tex
% Submitted to The Fibonacci Quarterly 20141106.
% Acknowlegement from Curtis Cooper (the editor) 20141117.
% Sent enquiry to Cooper 20150603
% Rejected - report received 20150720 - see Hadamard/minors/FQ/BOOP-report1.pdf
% Note that most of the open qns have been proved by Krattenthaler, 
% see ps/Krattenthaler/brent*.tex, also Bostan et al,
% and some results considerably generalised
% by Brent, Krattenthaler and Warnaar, so not clear if we want to submit
% elsewhere or not.
% 20151226: Maybe submit shortened version to an "easy" journal just to give
% Ohtsuka a publication and placate Prodinger ?
% See Hadamard/minors/BOOP-v4.*
% 20151228: Submitted shorter version to Hikari journal International
% Mathematical Forum.  See Hadamard/minors/Hikari/BOOP-IMF-v1.*
% (based on Hadamard/minors/BOOP-v4.* but with minor changes)
% 20151228: not considered by IMF as their backlog is too long!
% Tried Hikari's Pure Mathematical Sciences but turned down on the grounds
% that one of the references (Krattenthaler) was a personal communication!
% Maybe try Journal of Integer Sequences or Integers
% 20160131: submitted to Journal of Integer Sequences (JIS),
% see Hadamard/minors/JIS/temp/BOOP-JIS-v1.tex, email to Shallit
% also submitted the same (rpb260v2) to arXiv (appeared as arXiv v2)
% 20160202: Changed Thm numbers etc in OEIS entries to refer to arXiv v2.
% 20160201: email from Jeffrey Shallit - paper sent to referee,
% list of Latex points to fix up ***
% 20160301: email from Jeffrey Shallit - accepted subject to minor changes.
% 20160303: sent revision BOOP-JIS-v2.tex to Shallit (final version?)
% see Hadamard/minors/JIS/BOOP-JIS-v2.*
% 20160406: published in JIS, see email from Shallit and
% https://cs.uwaterloo.ca/journals/JIS/VOL19/Brent/brent9.pdf

\bibitem{rpb261}
\rpb, M. Coons % Michael Coons
and W. Zudilin, % Wadim Zudilin
Algebraic independence of Mahler functions via radial asymptotics,
% arXiv:1412.7906v1, 26 Dec.\ 2014. % Submitted by Wadim
% paper pwd e4nf7
% See ps/Zudilin/*, sources/arXiv/rpb261v1.tex, also Dropbox folder
\emph{International Mathematics Research Notices} 2016:2 (2016), 571--603,
% (accepted 22 April 2015).
% (published electronically 19 May 2015), 33~pp.
% http://imrn.oxfordjournals.org/content/early/2015/05/19/
% imrn.rnv139.full.pdf+html
\url{http:doi.org/10.1093/imrn/rnv139}.
MR3493426, % review by S. A. Stepanov
Also arXiv:1412.7906v2, 4 April 2015, 23~pp.
% Wadim submitted to International Mathematics Research Notices,
% see email 20150122, manuscript ID IMRN-2015-055
% pwd cjkwt25j -> Mahler261, user id Richard.Brent@anu.edu.au
% url https://mc.manuscriptcentral.com/imm
% Wadim is the corresponding author.
% Referees' reports received 20150401 - for the only one that
% said anything see ps/Zudilin/BCW.pdf
% Seems probable that it is accepted subject to minor revisions.
% Wadim submitted revision 20150404, ref. IMRN-2015-055.R1, see email.
% also new version on arXiv, see BCZ2015-04-04.*
% Accepted 20150422 (see email)

\bibitem{rpb262}
\rpb, J.\ H.\ Osborn and W.\ D.\ Smith,
Probabilistic lower bounds on maximal determinants of binary matrices,
\emph{Australasian Journal of Combinatorics} 66 (2016), 350--364.
% vol 66, number 3, accepted 27 July 2016, appeared on 31 Oct 2016.
Also
% 19 Jan.\ 2015,
% arXiv:1501.06235v4, 26 Jan.\ 2016.
%arXiv:1501.06235v5, 1 Aug.\ 2016, 17~pp.
%arXiv:1501.06235v6, 18 Aug.\ 2016, 17~pp.
% Following v7 has one more reference and slightly better text than
% the AJC version.
arXiv:1501.06235v7, 25 Oct.\ 2016, 17~pp.
% pwd xqnzz
For a longer version see \#257 above.
% Submitted D_bound12 = BOS_AJC_v1 to AJC 20150119
% reference number J138, see email from Elizabeth Billington 20150120
% arXiv/rpb262v1 = sources/arXiv/rpb262v1 submitted to arXiv 20150126
% arXiv/rpb262v2 = sources/arXiv/rpb262v2 submitted to arXiv 20150218
% arXiv/rpb262v3 = sources/arXiv/rpb262v3 submitted to arXiv 20150301
% arXiv/rpb262v4 = sources/arXiv/rpb262v4 submitted to arXiv 20160126
% Enquiry to AJC 20150803 - reply dated 20150804 (see emails)
% Referee's report received 20160113 - generally good but need to
% fix proof of Lemma 7.
% See Hadamard/minors/BOS_AJC_v2 = rpb262v4 for revision, 
% resubmitted to AJC 20160126.
% Receipt acknowledged by Elizabeth Billington 20160129.
% Email from Elizabeth Billington 20160727 - paper finally accepted!
% See Hadamard/papers/rpb262/ BOS_AJC_v3.tex for version with small typo 
% corrected 20160510 (final version is Hadamard/papers/rpb262/BOS_J138).
% Page proofs from Elizabeth Billington 20161024
% returned (2 corrections) and legal authority 20161025.
% Vol. 66(3), 2016, pp. xx-xx

\bibitem{rpb263}
\rpb, C.\ Krattenthaler and [S.] O.\ Warnaar,
Discrete analogues of Mehta-type integrals,
{\em J.\ Combin.\ Theory Ser.\ A} 144 (2016), 88--138,
\url{http://dx.doi.org/10.1016/j.jcta.2016.06.005}.
% Special issue ``Fifty Years of the Journal of Combinatorial Theory''.
% Published electronically 20160715, see rpb263-JCTA.pdf
MR3534065, % review by Timothy Y. Chow
Also arXiv 1601.06536, 25 Jan.\ 2016, 50~pp.
% Ole submitted to arXiv 20160122, submit/1461521, 1601.06536, pwd ihvg7
% Ole also submitted to special issue of JCTA
% See ps/Warnaar/BKW15.tex etc. (and also Hadamard/minors/BKW-summary.*)
% for my notes etc.).
% Two referees' reports received 20160502 - both good and paper accepted
% modulo minor corrections (to be done by Ole).
% 20160705: Proofs for YJCTA_4763 available, according to email from Ole.
% 20160706: Downloaded proofs, see ps/Warnaar/YJCTA_4763_proofs.pdf
% 20160715: Now visible on Elsevier website, see email from Ole.

\bibitem{rpb264}
\rpb, A.\ Kruppa and \pz,
FFT extension for algebraic-group factorization algorithms,
chapter in \emph{Topics in Computational Number Theory inspired by
Peter L.\ Montgomery}, J.\ Bos and A.\ Lenstra (editors),
Cambridge University Press, 189--205, 2017.
MR3753113. % Review by Peter Shiu
% Sent preliminary to Joppe Bos and Arjen Lenstra 20150718.
% Note: authors may include Bruce Dodson
% according to emails from Arjen and PZ 20160216,
% or maybe Alexander Kruppa (email from Arjen 20160217); 
% the editors may invite
% new authors (e.g. Kruppa, Kleinjung, Bos, Arjen L) for this chapter.
% Update: according to email 20160614, Dodson has been replaced by Alex Kruppa.
% Sent new signed contract to CUP 20160730
%% See ecmfft/chap10.tex [now chap9.tex], ecmfft/book.pdf, svn/notes etc.
% 20161230: See Petmon_FFT_ext.pdf for version updated by Alex Kruppa.
% Final version 20170131 (or later - see emails from Joppe Bos 20170215).
% Expected to be published 20171012, http://www.cambridge.org/9781107109353
% Preliminary versions in ecmfft/*
% Almost final version www/pd/rpb264-Ch8.pdf (from Joppe Bos 20170926)
% Final version  ecmfft/book_ch8.*, www/pd/rpb264.pdf (from Joppe Bos 20181110).
% ISBN 9781107109353, price AUD195(!)
Also \url{https://hal.inria.fr/hal-01630907}.
% rpb264-HAL.pdf from PZ 20171109

\bibitem{rpb265}
David H. Bailey, Jonathan M. Borwein, Richard Brent, 
and Mohsen Reisi Ardali,
Reproducibility in computational science: a case study:
randomness of the digits of $\pi$,
{\em Experimental Mathematics} 26 (2017), 298--305. % issue 3
Published online 24 Aug.\ 2016,
\url{http://dx.doi.org/10.1080/10586458.2016.1163755}.
MR3642107. % Reviewed by Samuel S. Wagstaff Jr
% Submitted to Experimental Mathematics (by Jon Borwein).
% Manuscript ID UEXM-2016-0002
%
% Site URL:  https://mc.manuscriptcentral.com/experimentalmath
% User ID:    Richard.Brent@anu.edu.au
% Your password is: https://mc.manuscriptcentral.com/experimentalmath?URL_MASK=4202f0dd7ce94aceacf8acca7c38c9c3
%
%See pi-repro.pdf,  emails 20150806..20160103.
% Report (by Ganz) received 20160222, see ps/Borwein/UEXM-2016-0002_report.pdf
% outcome is unclear.
% Accepted - see email from Borwein/Tabachnikov 20160307.
% See pi-repro/pi-repro-v08.* for latest version (20160308)
% See ps/Borwein/UEXM*, rpb265-proofs.pdf for page proofs etc. as at 20160816
% Appeared 20160824, see emails

\bibitem{rpb266}
\rpb\ and \pz, % Richard P. Brent & Paul Zimmermann
Twelve new primitive binary trinomials,
arXiv:1605.09213v1, 24 May 2016, 2~pp.
% Submitted to Math. Comp. by PZ, 20160509, Id 160509-Zimmermann
% but rejected as unsuitable (see email from PZ 20160519).
% Announced at
% https://listserv.nodak.edu/cgi-bin/wa.exe?A2=ind1605&L=NMBRTHRY&F=&S=&P=50
% Submit to arXiv and don't try other journals.
% Submitted to arXiv 20160524, submit/1568339.  pwd yuwfh
% See sources/rpb266.tex

\bibitem{rpb267}
\rpb,
Asymptotic approximation of central binomial coefficients with
rigorous error bounds,
\emph{Open J.\ of Mathematical Sciences}, to appear.  
% can abbreviate as \emph{Open J.\ Math.\ Sci.}
Also
% arXiv:1608.04834v1, 17 August 2016, 11~pp.
% arXiv:1608.04834v2, 15 Sept.\ 2016, 11~pp.
% arXiv:1608.04834v3, 13 April 2021, 11~pp.
arXiv:1608.04834v4, 15 Sept.\ 2021, 11~pp.
% arXiv pwd 5rykh
%
% 20210413: Submitted to Open Journal of Mathematical Sciences (O[J]MS),
% see email from Waqas Nazeer 20210319 and my reply of 20210413.
% Received acknowledgement 20210414, submission ID is OMS278034,
% username rpbrent@gmail.com, see http://scholarmanuscript.com
% pwd saved by Firefox, submission currently being processed (at 20210414)
%
% See Hadamard/minors/central_binomial.* for early version,
% and sources/rpb267-OJMS.tex for version submitted to OJMS.
% If OJMS turns it down, should try JIS (Shallit).
% 20210813: Sent reminder to editor via online system.
% 20210824: Sent email reminder to editor as no reply to previous message.
% 20210913: Reply from editor Waqas Nazeer - he sent one referee's report
% (mainly favourable) and is expecting another one soon.
% See Hadamard/minors/rpb267-OJMS.tex, rpb267-ref1.txt
% and rpb267v4.tex for current draft of final revision.
% 20210915: Paper accepted by OJMS subject to some revisions, see email
% See Hadamard/minors/rpb267v4 for new arXiv version,
% rpb267-OJMS for OJMS version (the same except for bib style),
% and rpb267-response-to-referee.txt for response to the referee.
% 20210915: Submitted rpb267-OJMS.pdf (not yet .tex) and my response to
% the referee's comments rpb267-response-to-referee.txt
% 20210915: Received acknowledgement, new id is OMS278034_R1.

\bibitem{rpb268}
\rpb,
On asymptotic approximations to the log-Gamma and \hbox{Riemann-Siegel}
theta functions,
\emph{J.\ Aust.\ Math.\ Soc.,} 107 (2019), 319--337.
% published online 21 Dec.\ 2018.
% DOI: https://doi.org/10.1017/\\
% S1446788718000393.
MR4034593.   % Review by Richard Paris
Also arXiv:1609.03682, 6 Oct.\ 2016, 23~pp.
% arXiv v1 = RS-theta2.tex, 20160913
%  there is a typo in Thm 2, page 9, k-1/2 should be k+1/2 (fixed in v2)
% arXiv v2 = RS-theta3.tex, 20161006
% arXiv pwd vpbmj
% Submitted to Jon Borwein's memorial J. AustMS volume 20170712,
% ref. JAZ-RA-17-0140. See Borwein/JAMS/notes.txt
% Riemann/RS-theta[2-3].tex, Borwein/JAMS/JAMS-Borwein-Brent_v1.*
% Asked Brailey Sims 20180202 if there is any news about it.
% He replied that it has been received and is being refereed.
% Sent reminder to Brailey 20180806.  He will look into it.
% 20181011: paper accepted by the editor (George Willis).
% Officially received 20170712, accepted 20181010.
% Need to return final version (v2) and copyright form (done).
% Publisher website https://mc.manuscriptcentral.com/jaz
% Username rpbrent@gmail.com, pwd r..jams12
% New version (v2) JAZ-RA-17-0140.R1 submitted via website 20181018,
% also sent copyright form via email.  Don't submit new arXiv version
% as no significant mathematical changes.
% 20181113: received pdf of page proofs - returned 20181114
% (suggested change in footnote on page 3 re new ref [6]).
% Published, for access, cambridge@rpbrent.com, i..c.Cambridge1

\bibitem{rpb269}		% See also rpb273
\rpb,
The Borwein brothers, $\pi$ and the AGM,
in~\cite{rpb273}, 323--348.
%\emph{From Analysis to Visualisation: A Celebration of the Life and Legacy
%of Jonathan M.~Borwein
%Proceedings of the Jonathan Borwein Commemorative Conference % JBCC
%}
%(edited by Brailey Sims \emph{et al}),
%%(edited by D. H. Bailey, N. Borwein, R. P. Brent, R. S. Burachik, 
%%J. H. Osborn, B. Sims and Q. J. Zhu)
%Springer Proceedings in Mathematics and Statistics, Vol.\ 313, 
%2020, 323--348.
% ISSN 2194-1009, 2194-1017 (electronic)
% ISBN 978-3-030-36567-7, 978-3-030-36568-4 (eBook)
% https://doi.org/10.1007/978-3-030-36568-4_21
% JBCC 2017 (Springer Proceedings in Mathematics and Statistics 313)
% Published March 2020, see 
% https://link.springer.com/book/10.1007/978-3-030-36568-4
Also arXiv:1802.07558, 8~Aug.\ 2018, 24 pp. % v2
% Also arXiv:1802.07558v1, 21 Feb.\ 2018, 24 pp.
% arXiv:1802.07558, rpb269v1, pwd 42xks
% See Borwein/JBCC/editorial/Brent/JBCC-RPB-proc.*
% Submitted to Brailey Sims 20180221 for JBCC Proceedings
% Sent reminder to Brailey 20180806.
% Reply from Brailey 20180807 - accepted, no changes required
% but for v2 I made some minor formatting changes and used the new 
% style file.
% Email from Springer editor suggests it will be published early 2020.
% Brailey received page proofs 20200331

\bibitem{rpb270}
\rpb\ and A.\ B. Yedidia,	% Adam B. Yedidia
Computation of maximal determinants of binary circulant matrices,
\emph{Journal of Integer Sequences} 21 (2018), article 18.5.6, 19 pp.
MR3830901.
% https://cs.uwaterloo.ca/journals/JIS/VOL21/Brent/brent11.pdf
% arXiv:1801.00399v1, 31 Dec.\ 2017, 18 pp.	% pwd fizhp
% arXiv:1801.00399v2, 10 Jan.\ 2018, 18 pp.
% arXiv:1801.00399v3, 19 Apr.\ 2018, 22 pp.
% arXiv:1801.00399v4, 6 June 2018, 22 pp.
% arXiv:1801.08399v5, 11 June 2019, 22 pp.
% arXiv:1801.00399v6, 20 Feb 2021, 22 pp.
Also (updated version)  arXiv:1801.08399v6, 20 Feb 2021, 22 pp.
% submit/2118419, 6pm Aust. time 20171231.
% resubmitted 5:30pm Aust. time 20180101 with fix to Table 3.
% Submitted v2 20180110 (better notation and extended tables).
% Submitted v3 20180419 (more counter-examples and extended tables).
% Submitted v4 20180606 (corrections and further explanations).
% Submitted v6 20210217 (now goes to $n=53$)
% See circulants/maxdet-circulants.tex, sources/rpb270.tex
% Submmitted 20180425 to Journal of Integer Sequences (Shallit),
% for this version see circulants/JIS/maxdet-circ-JIS-v1.*
% Referee's report received 20180601 - need to make a few changes,
% but seems to be accepted in principle.
% Submitted circulants/JIS/maxdet-circ-JIS-v2.* (with comments removed) 
% to JIS (Shallit) 20180606 (accepted)
% appeared in JIS 20180610.
% Postscript (20190611): see circulants/JIS/v5-comments.tex
% for revision with an update of Table 2 (including entry for n=53)
% and corresponding sources/arXiv/rpb270v5.
% Later update Tabe 3 for n=53					***

\bibitem{rpb271}
\rpb, A. J. Guttmann and M. L. Glasser,
A conjectured integer sequence arising from the exponential integral,
\emph{Journal of Integer Sequences} 22 (2019), article 19.4.7, 16 pp.
MR4001007.
% Published 15 August 2019.
% arXiv:1812.00316v1, 2 Dec.\ 2018. 
% arXiv:1812.00316v2, 15 Dec.\ 2018. 
% arXiv:1812.00316v3, 27 May 2019. 
Also arXiv:1812.00316v4, 31 May 2019. 
% See ps/Guttmann/BGG3.tex, 20181201
% Submitted to arXiv 20181102, arXiv:1812.00316v1, pwd rwdzf
% see sources/arXiv/rpb271v1.tex
% Version ps/Guttmann/JIS/BGG-JIS1.tex submitted to JIS (Shallit) 20181214,
% this is the same as ps/Guttmann/BGG4.tex except for comments removed.
% Corrected defn of $E_1$ on page 2 in version sent 20181215.
% Shallit acknowledged 20181218.
% See email 20190520 from Shallit - accepted subject to minor changes.
% Returned new version ps/Guttmann/JIS/BGG-JIS2.tex to Shallit 20190527
% Expected to appear by 20190901 - see email from Shallit 20190806 ***
% Also submitted to arXiv, 1812.00316v1, 20181202
% and 1812.00316v2, 20181214, corrected 20181215.
% and 1812.00316v3, 20190527 (added motivation and some references),
% and 1812.00316v4, 20190531 (added AJG acknowledgement).
% See sources/arXiv/rpb271v[2-4].tex, ps/Guttmann/JIS/BGG-JIS*.tex

\bibitem{rpb272}
\rpb, C.~Pomerance, D.~Purdum, and J.~Webster,
% \rpb, Carl Pomerance, David Purdum, and Jonathan Webster,
Algorithms for the multiplication table problem,
\emph{Integers}, to appear. % Accepted 20210812, see below.
% arXiv:1908.04251v1, 12 August 2019, 14~pp.
Also arXiv:1908.04251v2, 5 May 2021, 15~pp.
% See ps/Webster/Multiplication\_local/erdos_mult_table_v*.*
% renamed to ps/Webster/Multiplication/joint_papers/erdos_mult_table_v*.*
% JW submitted a slightly modified version to ANTS 20200222,
% paper ID: 200221-Webster
% paper submission deadline 20200225 (met),
% acceptance/rejection notification 20200427,
% see https://www.math.auckland.ac.nz/~sgal018/ANTS/callforpapers.html
% 20200428: email from Steven Galbraith (via Jonathan) - paper was rejected.
% Waiting for more details and/or referees' reports
% Carl suggests trying Experimental Mathematics or Math Comp.
% 20210414: sent email to the three co-authors suggesting that we
% try Experimental Mathematics.
% 20210505: RPB submitted v10a = erdos_mult_table_arXiv_v2 to arXiv.
% 20210506: Jonathan submitted v10a to Experimental Mathematics,
% see emails.  Reference number is 215805387.
% 20210607: Rejected by Experimental Mathematics, see emails.
% 20210617: Submitted to Integers, submission #3362, see email from Jonathan
% 20210621: Changed directory, files now in ps/Webster/joint_papers/
% 20210812: Accepted by Integers - see email from Bruce Landman via Jonathan
% 20210820: Jonathan sent the revised version (i.e. v12) to Landman.
%	    Paper will probably appear in Vol. 21 (2021) or 22 (2022).
%	    See http://math.colgate.edu/~integers/ for contents etc.

\bibitem{rpb273} %  JBCC Proceedings, see also rpb269
D. H. Bailey, N. S. Borwein, \rpb, R. S. Burachik,
J. H. Osborn, B. Sims, and Q. J. Zhu (editors),
% David H. Bailey, Naomi Simone Borwein, \rpb, Regina S. Burachik,
% Judy-anne Heather Osborn, Brailey Sims, and Qiji J. Zhu (editors),
\emph{From Analysis to Visualisation: A Celebration of the Life and Legacy
of Jonathan M.~Borwein, Callaghan, Australia, September 2017}.
Springer Proceedings in Mathematics and Statistics, Vol.\ 313, 
2020, xxi~$+$~439~pp.
ISSN 2194-1009, 2194-1017 (electronic),
ISBN 978-3-030-36567-7, 978-3-030-36568-4 (eBook).
\url{https://doi.org/10.1007/978-3-030-36568-4}
% Published March 2020, see 
% https://link.springer.com/book/10.1007/978-3-030-36568-4
% A complete electronic copy of the book is at ps/Borwein/JBCC-book.pdf
% To login, richard.brent@anu.edu.au, i..c.Springer1

\bibitem{rpb274}
\rpb, D. J. Platt, and T. S. Trudgian,
%\rpb, David J. Platt, and Timothy S. Trudgian,
The mean square of the error term in the prime number theorem,
\emph{J.\ Number Theory}, to appear. 
% See https://www.journals.elsevier.com/journal-of-number-theory
Also 
arXiv:2008.06140v1, 14 August 2020, 23~pp.
% Date in arXiv version is 17 Aug, but in margin it is 14 Aug.
% arXiv pwd detr9
%
% 30300813:  Submitted to IMRN by Tim Trudgian, ms id IMRN-2020-815.
% Site URL: https://mc.manuscriptcentral.com/imrn
% USER ID:  Pintz@rpbrent.com
% pwd: IMRN585810
% 20200821: rejected by IMRN as "too specialised" though referee's report
% mildly encouraging.
%
% 20200827: Submitted to J. Lond. Math. Soc. by Tim Trudgian (Adam Harper
% editor). Paper ID: 200827-Trudgian.
% To track status, go to 
% https://edf.lms.ac.uk/status.php?p_id=37511&cr=95F963EC3C
% 20210224: rejected, suggested using a more specialist journal.
% Also pointed to a possible error in one of the references which was used,
% see ps/Bennett/counting_zeros.pdf
% See ps/Trudgian/joint_papers/BPTCv*, Brent_Platt_Trudgian_Cramer_submitted.*
% This incorporates most of omega_result6 and closer_to_Pintz8
% Check title of Stechking and Popov paper in next revision.
%
% 20210316: TT submitted to Algebra and Number Theory, suggesting
% Editor: Andrew Granville, Paper ID: 210315-Trudgian 
% You can track the status of your article here:
%  https://ef.msp.org/status.php?p_id=110660&cr=CB5A1EA7E8
% 20210406: Granville turned it down for Algebra and Number Theory
%
% 20210416: TT submitted to Math. Comp., 
% 20210517: Rejected by Math. Comp., see letter from Susanne Brenner
%
% 20210518: Tim submitted [5th attempt] to JNT with minor changes to references
% 20210727: Conditionally accepted for JNT (see email from Tim),
%           just have to make some minor changes to satisfy the referee
% 20210831: Tim returned to JNT with minor changes and new log-scale plot

\bibitem{rpb275}
\rpb, D. J. Platt, and T. S. Trudgian,
%\rpb, David J. Platt, and Timothy S. Trudgian,
On a harmonic sum over non-trivial zeros of the Riemann zeta-function,
\emph{Bull.\ Austral.\ Math.\ Soc.\ }104 (2021), 
59--65. % Issue 1, August 2021.
% Submitted by Tim Trudgian to Bull. AustMS about 20200911.
% see ps/Trudgian/notes_on_Hassani*.tex
% Accepted 20200914 by John Loxton for publication in Bull. AustMS.
% 20200916: Tim Trudgian sent the tex version to John Loxton.
% Received proofs, see ps/Trudgian/BAZ_2000125_PRF.pdf,
% told TT that they were OK  20201021.
% published online 20 Nov.\ 2020.
% \url{https://doi.org/10.1017/S0004972720001252}.
% Published online on or before 20210314, see email.
% MR4279314 (Pending)
Also arXiv:2009.05251v1, 11 Sept.\ 2020, 8~pp.
% arXiv paper pwd gbmbu

\bibitem{rpb276}
\rpb, D. J. Platt, and T. S. Trudgian,
%\rpb, David J. Platt, and Timothy S. Trudgian,
Accurate estimation of sums over zeros of the Riemann
zeta-function, \MC\ 90 (2021), 2923--2935.
% Vol 90, Number 332, November 2021, published online 23 June, 2021.
\url{https://doi.org/10.1090/mcom/3652}. 
Also arXiv:2009.13791v1, 29 Sept.\ 2020, 15~pp.
% TT submitted it to Math Comp 20200929, 
% and also to arXiv (pwd iybz2).
% 20210301: received two referees' reports from the editor (Susanne C. 
% Brenner); looks like the paper will be accepted after minor revisions. 
% 20210310: Tim sent back revision BPT_Accel_Post_v5 and responses.
% 20210325: Accepted - see email from Susanne Brenner.
% 20210328: RPB put in Math Comp format, TT returned to Math Comp.
% 20210331: RPB signed copyright form for MCOM3652 (other authors need
%	    to sign independently).
% 20210604: Now visible under ``early view'' at the Math Comp website,
%	    DOI: https://doi.org/10.1090/mcom/3652
%	    (although I was unable to login to my AMS account when I tried)
% 20210621: Tim returned the page proofs.
% 20210623: Published electronically, access via DOI.
% See ps/Trudgian/joint_papers/BPT_Accel_Post_v6.*

\bibitem{rpb277}
\rpb,
On some results of Ag\'elas concerning the GRH and of Vassilev-Missana
concerning the prime zeta function,
arXiv:2103.09418v2, 21 March, 2021, 6~pp. 
% v1 appeared on arXiv 18 March 2021, dated 17 March 2021.
% v2 appeared on arXiv 23 March 2021, dated 21 March 2021.
% The paper password for arXiv is: kxwq4
% For a shorter version, see rpb278. Used journal ref for it.
% For a (much) longer version, see rpb279.

\bibitem{rpb278}
\rpb,
On two theorems of Vassilev-Missana, 
\emph{Notes on Number Theory and Discrete Mathematics}
{27}, 2 (2021), 49--50.
% appeared on or before 25 June 2021 (accepted 1 June 2021).
% This is a shorter version of rpb277.
% 20210323: submitted to Notes on Number Theory and Discrete Mathematics.
% Received acknowledgement from Vassia Atanassova 20210323,
% reference ID 2021/271.
% 20210601: Accepted (see email from Vassia Atanassova,
% expected to appear in NNTDM 27/2. Appeared.
% She also sent me an erratum by Vassilev-Missana to referee - done).
% 20210604: Returned galley proofs.
% Details are Vol. 27, 2021, No. 2, XX-XX
% DOI: 10.7546/nntdm.2021.27.2.49-50
% Received 23 March 2021, accepted 1 June 2021
% Note - this version is not on arXiv, but the longer version rpb277 is,
% so added the journal reference to the info for the arXiv version.
% See ~/GRH/* for assorted files.

\bibitem{rpb279}
\rpb,
Some instructive mathematical errors,
\emph{Maple Transactions} {1}, 1 (2021), 4.1--4.16.
% Appeared 24 June 2021 (accepted 14 June 2021).
Also arXiv:2106.07269v2, 20 June 2021, 25~pp.
% Greatly extended version of rpb277 (which is now Sec. 3).
% 20210613: Submitted to Rob Corless for Maple Transactions (a new journal).
% 20210614: Accepted. Problem with ORCID fixed 20210623.
% 20210614: Also arXiv (v1) % pwd wyht5
% 20210620: Also arXiv (v2)
% See  ~/GRH/Brent-Corless*.*, rpb279*

\bibitem{rpb280} 
\rpb,
Jonathan M. Borwein 1951--2016: Life and Legacy,
\emph{Maple Transactions} 1, 2 (2021), % November 2021 issue
to appear.
Also 
% arXiv:2107.06030v1, 13 July 2021, 18 pp.	% pwd nr4xh
% arXiv:2107.06030v2, 18 July 2021, 18 pp.	% pwd nr4xh
%arXiv:2107.06030v3, 21 Aug 2021, 19 pp.	% pwd nr4xh
arXiv:2107.06030v4, 15 Sept 2021, 19 pp.	% pwd nr4xh
% 20 May 1951 -- 2 August 2016,
% \emph{Historical Records of Australian Science}, submitted but withdrawn.
% Note that the HRAS version has a slightly different title.
% See emails August 2016 for invitation etc.
% Note (20200722): emails to/from Chris Dickman
% who is now the editor. He says we can use Latex, and OK to ask Brailey
% Sims to be a co-author instead of David Bailey
% (but in the end I didn't ask either of them).
% 20210713: Submitted JMB-obit-v1 to HRAS (Chris Dickman, editor)
% See Borwein/obituary/JMB-obit-v1 etc, sources/rpb280.tex etc.
% 20210717: submitted JMB-obit-v1b.pdf to HRAS via ScholarOne,
% Manuscript ID is HR21006.
% See Borwein/obit/JMB-obit-v1b.*
% 20210717/18: submitted rpb280v2 to arXiv (fixed some typos, new abstract)
% 20210820: accepted subject to "minor" revisions - the text is OK but
% the editor wants some unnecessary and time-consuming changes in the format.
% Note - the referees' reports, possibly by Tony Guttmann and Brailey Sims,
% were favourable.
% 20210821: Withdrew ms from consideration by HRAS (see email to Ian Rae).
% Also informed Karen Holt at the Academy (bcc Tony Guttmann, Brailey Sims).
% 20210821: Submitted anonymised version rpb280v3anon to Maple Transactions,
% see email to Rob Corless. First referee favorable, awaiting second report.
% 20210910: Paper accepted (second referee had no useful comments).
% Now need to add a reference as suggested by first referee, and change	***
% to Maple Transactions format.
% For latest versions, see Borwein/obit/rpb280v4 (plain style)
% and Borwein/obit/Maple_Trans/rpb280-MT (Maple Transactions style).
% 20210913: Sent revision rpb280-MT and associated files to Rob Corless.

%\bibitem{rpb28x}
% Juan Arias de Reyna, \rpb, and Jan van de Lune,
% In preparation
% See ../Riemann/log-zeta[-v8].tex for ABL "log-zeta" paper, 
% Started 20120526, stopped 20120706, still in draft form.

%\bibitem{rpb28x}
% \rpb\ and J.\ H.\ Osborn,
% Fast computation of the Kolakoski summatory function,
% in preparation.
% Submit to JIS?  See OEIS A195206 etc.

%\bibitem{rpb28x}
% Brent, Krattenthaler and Warnaar,
% Non-intersecting lattice paths and identities for classical group
% characters, in preparation. (Based on Krattenthaler's talk at the
% Guttmann 70 conference?)

%\bibitem{rpb28x}
% A paper that I promised to write on Jon Borwein and
% Experimental Mathematics - not started yet.
% See email from Bharath Sriraman 20200616 etc.			**
% Note the 2010 paper by Bailey and Borwein in Notices AMS,
% may have useful material.

\bigskip
\rightline{Last revised 16 September 2021}
\vspace*{\fill}

% Note: for Google scholar citations see
% http://scholar.google.co.nz/citations?user=TRcAF4cAAAAJ&hl=en

\end{thebibliography}
\end{document}
